Principles for the design of multicellular engineered living systems 🕫 📾

Cite as: APL Bioeng. **6**, 010903 (2022); doi: 10.1063/5.0076635 Submitted: 26 October 2021 · Accepted: 2 February 2022 · Published Online: 2 March 2022



Onur Aydin,^{1,2} D Austin P. Passaro,³ D Ritu Raman,⁴ Samantha E. Spellicy,^{3,5} Robert P. Weinberg,⁶ Roger D. Kamm,^{7,8} D Matthew Sample,⁹ D George A. Truskey,¹⁰ D Jeremiah Zartman,¹¹ Roy D. Dar,¹² Sebastian Palacios,¹³ Jason Wang,¹² Jesse Tordoff,¹⁴ Nuria Montserrat,¹⁵ Rashid Bashir,^{12,16} M. Taher A. Saif,¹ and Ron Weiss^{8,13,a})

AFFILIATIONS

- ¹Department of Mechanical Science and Engineering, University of Illinois at Urbana-Champaign, Urbana, Illinois 61801, USA ²Carl R. Woese Institute for Genomic Biology, University of Illinois at Urbana-Champaign, Urbana, Illinois 61801, USA
- ³Regenerative Bioscience Center, University of Georgia, Athens, Georgia 30602, USA
- ⁴Department of Chemical Engineering, Massachusetts Institute of Technology, Cambridge, Massachusetts 02139, USA
- ⁵University System of Georgia MD/PhD Program, Medical College of Georgia at Augusta University, Augusta, Georgia 30912, USA
- ⁶School of Pharmacy, Massachusetts College of Pharmacy and Health Sciences, Boston, Massachusetts 02115, USA
- ⁷Department of Mechanical Engineering, Massachusetts Institute of Technology, Cambridge, Massachusetts 02139, USA
- ⁸Department of Biological Engineering, Massachusetts Institute of Technology, Cambridge, Massachusetts 02139, USA
- ⁹Center for Ethics and Law in the Life Sciences, Leibniz Universität Hannover, 30167 Hannover, Germany
- ¹⁰Department of Biomedical Engineering, Duke University, Durham, North Carolina 27708, USA
- ¹¹Department of Chemical and Biomolecular Engineering, University of Notre Dame, Notre Dame, Indiana 46556, USA
- ¹²Department of Bioengineering, University of Illinois at Urbana-Champaign, Urbana, Illinois 61801, USA
- ¹³Department of Electrical Engineering and Computer Science, Massachusetts Institute of Technology, Cambridge, Massachusetts, 02139, USA
- ¹⁴Computational and Systems Biology Program, Massachusetts Institute of Technology, Cambridge, Massachusetts 02139, USA
- ¹⁵Institute for Bioengineering of Catalonia (IBEC), The Barcelona Institute of Science and Technology (BIST), 08028 Barcelona, Spain
- ¹⁶Department of Electrical and Computer Engineering, University of Illinois at Urbana-Champaign, Urbana, Illinois 61801, USA

^{a)}Author to whom correspondence should be addressed: rweiss@mit.edu

ABSTRACT

Remarkable progress in bioengineering over the past two decades has enabled the formulation of fundamental design principles for a variety of medical and non-medical applications. These advancements have laid the foundation for building multicellular engineered living systems (M-CELS) from biological parts, forming functional modules integrated into living machines. These cognizant design principles for living systems encompass novel genetic circuit manipulation, self-assembly, cell–cell/matrix communication, and artificial tissues/organs enabled through systems biology, bioinformatics, computational biology, genetic engineering, and microfluidics. Here, we introduce design principles and a blueprint for forward production of robust and standardized M-CELS, which may undergo variable reiterations through the classic design-build-test-debug cycle. This Review provides practical and theoretical frameworks to forward-design, control, and optimize novel M-CELS. Potential applications include biopharmaceuticals, bioreactor factories, biofuels, environmental bioremediation, cellular computing, biohybrid digital technology, and experimental investigations into mechanisms of multicellular organisms normally hidden inside the "black box" of living cells.

© 2022 Author(s). All article content, except where otherwise noted, is licensed under a Creative Commons Attribution (CC BY) license (http:// creativecommons.org/licenses/by/4.0/). https://doi.org/10.1063/5.0076635

I. INTRODUCTION

It is now possible to create novel multicellular living machines never seen before. The past few decades have witnessed the convergence of the fields of synthetic biology, bioengineering, stem cell biology, computational biology, and molecular genetics, coupled with increased training of interdisciplinary scientists across two or more of these domains (Fig. 1).¹⁻⁴⁶ This convergence has been accompanied by advances in the design and implementation of technologies such as organoids, microfluidics, biological robots (biobots), nanofabrication, and genetic engineering. The specialized knowledge gained from these interdisciplinary studies now enables the design and engineering of "multi-cellular engineered living systems" (M-CELS) through the combination of all of these technologies. M-CELS represent an engineering approach to building living machines and devices using biological building blocks. They utilize classic engineering modalities of design, modeling, prototype fabrication, testing, and iteration but draw from a toolbox that includes living cells, organelles, proteins, nucleic acids, and biomaterials. The two basic objectives of building living cellular machines are (1) to better understand the properties, mechanisms, and dynamics of these parts in normal living organisms ("to build is to understand")⁴⁷ and (2) to create better, more intelligent, and biologically efficient machines that can dynamically adapt to their environments in a manner not technically feasible with abiotic machines.

Designing living systems from the ground up requires developing a better understanding of biological building blocks and constructing a set of design principles that define how the blocks fit together and can be manipulated. These design principles enable M-CELS to bring a new perspective to how we think about biology. This fundamental understanding allows M-CELS to be developed for a range of societal applications, including regenerative medicine, agriculture, disease modeling, and pharmaceutics, but these applications currently remain highly challenging experimentally. Biobots (biological robots), for example, have been promoted for use in biomedical applications and pharmacology for developing platforms to gain insights into neural networks and for lightweight compliant actuation using living muscle⁴⁸ and organoids for disease modeling.⁴⁹ These anticipated applications should not be taken for granted but developed in careful engagement with the stakeholders or intended beneficiaries of the particular technology.⁵⁰ The design cycle, in this way, integrates technical, societal, and ethical perspectives and principles, in keeping with the transformative aspirations of the M-CELS field.

Our high-level definition of M-CELS is that these are engineered multicellular systems that have emergent behaviors with desired natural or non-natural form and/or function. This Review provides bioengineers with a set of primary design principles to guide the future design and manufacture of M-CELS. Here, we trace these design principles across three representative M-CELS-(1) organoids, (2) microphysiological systems, and (3) biobots. There are intrinsic and dynamic properties of multicellular systems, which must guide these principles a priori. These include spatial and gradient properties that play important roles in developmental biology and create the desired form and function. Such spatial cues guide the development of multicellular organisms by regulating cellular organization and fate, providing the benefit of improved productivity through specialization for organ function. As such, M-CELS are valuable systems for the study of complex emergent behaviors, including intercellular interactions and dynamics. Emergent behavior can be defined as the self-directed,

multicellular response occurring as a result of the collective interactions of individual cells between themselves and the extracellular environment that manifests itself by phenomena at the macroscopic systems-level scale.

M-CELS are characterized phenotypically by their overall complexity of form and function and share certain common properties including:

- (1) multicellular organization composed of cells with specialized types, functions, and stages of differentiation and maturation;
- (2) spatial organization generated through autonomous cell sorting, artificial instructions encoded by bio-macromolecules, and thermodynamically favorable self-assembly;
- (3) existence of multiple states characterized by different state variables with various degrees of freedom;
- (4) multiple modalities of cell-to-cell interactions, such as, but not limited to, specialized cell-cell junctions, ion channels, humoral receptors, electrochemical channels, cell-cell adhesion molecules, G-protein coupled receptors, and membrane-embedded transducing elements;
- (5) macroscale functional modalities (for example, long range contraction, perfusion, metabolic activities, shear fluidic flow, environmental sensing of heat, nutrients, toxins, danger signals, and gravitational and electromagnetic fields).

We present a discussion of the principles for M-CELS' design following an outline based on the typical stages of progression of M-CELS from development to maintenance and adaptation to function. Section II covers the early stages of biological development, focusing on spatiotemporal aspects of cell differentiation, pattern formation via self-organization, stochastic decision-making, and control of cell microenvironment. Section III discusses developmental processes related to the integration of multiple cellular modules (e.g., tissue constructs or organoids) and formation of interfaces, highlighting the roles of reciprocal inductive interactions, compartmentalization and timing, nutrient exchange and transport, and interfacing with abiotic materials. Section IV focuses on maintenance and monitoring of M-CELS, processes of degradation and regeneration, and adaptative responses to external cues and changes in environmental conditions. Section V discusses principles related to the functions of M-CELS, including sensing, information processing, and several modes of actuation (mechanical, chemical, optical, bioelectric, and structural). Finally, in Sec. VI, we present a summary of the framework of design principles presented in Secs. I-V and conclude with an outlook on the future of M-CELS highlighting the need for community-driven efforts and engagement to address potential technical, ethical, and societal challenges.

II. DEVELOPMENT PART I: EMBRYONIC DEVELOPMENT AND CELL DIFFERENTIATION

To articulate design principles for developing M-CELS, we consider a two-level hierarchical organization framework (Table I). The first level comprises individual subsystems (or modules) whose development is largely self-regulated (that is, these systems do not require feedback from other subsystems to attain primary form and function). The second level involves systems that develop through the integration of multiple modules where inductive interactions between the modules



FIG. 1. Evolutionary timeline of synthetic biology lying the foundation for M-CELS. (1962) Frog nuclei from intestinal somatic cells are re-programmed to recapitulate embryogenesis into adult frog,¹ (1964) cracking the triplet codon genetic code;² and (1970) discovery of DNA slicing/dicing with restriction endonucleases;³ laid foundation for recombinant DNA and genetic editing (1975) high-performance DNA sequencing;⁴ along with (1978) site-directed mutagenesis provided tools for fine dissection of gene function analysis;⁵ (1979) tissue-like architecture composed of fibroblast-synthesized collagen lattices provided framework for artificial organs;⁶ (1980) DNA cloning resulted in recombinant DNA and the birth of the biotechnology industry;⁷ (1982) discovery of the lambda phage lytic-lysogeny switch generated study of biologic circuits and systems biology,8 (1983) PCR amplification yielded working quantities of DNA for genomics from picograms; (1995) proteins were used as computational elements in living cells;⁹ (1995) electric circuits simulate genetic networks through hybrid modeling;¹⁰ (1997) artificial skin used to treat severe burn victims;¹¹ (1997) Dolly, the sheep, was first mammal cloned from adult somatic cell through nuclear transfer;¹² (1998) RNAi discovered as a tool for selective gene expression;¹³ (2000) genetic toggle switch designed in E. coli;¹⁴ synthetic oscillatory network designed utilizing cellular transcriptional regulators,¹⁵ (2001) cell-cell communication effectuates quorum sensing;¹⁶ (2001) Human Genome Project produces first map of the human genome; (2002) engineered gene circuits amenable to mathematical modeling;¹⁸ (2002) engineering and device physics of cellular logic gates;¹⁹ (2002) combinatorial synthesis of genetic circuits facilitates quantitative analysis of modules and systems;²⁰ (2002) chemical synthesis of poliovirus cDNA de novo;²¹ (2003) design of genetic circuitry demonstrating oscillatory or toggle switch behavior;²² (2003) idempotent vector design for standard assembly of BioBricks;²³ (2003) engineering the mevalonate pathway for production of terpenoids;²⁴ (2004) population control circuit effected through cell-cell communication and quorum sensing;²⁵ (2004) design of riboregulators facilitate posttranscriptional regulation of gene expression; (2005) genetically engineered multicellular pattern formation;²⁶ (2005) design of light-sensing E. coli;²⁷ (2006) induction of mouse iPSCs into specialized cells via defined factors;²⁸ (2007) universal RNAi-based logic evaluator in mammalian cells,²⁹ (2007) tunable mammalian genetic switch coupling repressor proteins with an RNAi target design,³⁰ (2008) tunable synthetic gene oscillator,³¹ (2008) synthetic RNA devices for higher-order cellular information processing;³² (2009) use of multiplex genome engineering and accelerated evolution for programming cells;³³ (2009) synthetic bacterial edge detection program;³⁴ (2009) synthetic gene networks which count numerically;¹⁷ (2010) control of bacterial cell by genome chemically synthesized de novo;^{35,36} (2011) differentiation of functional human hepatocytes from iPSCs;³⁷ (2011) multicellular computation via genetically encoded NOR gates;³⁸ (2011) stripe patterns produced by synthetic genetic circuit;³⁹ (2011) metabolic genetic switchboard designed on riboregulators,⁴⁰ (2011) creation of sensing array from coupled biopixels;41 (2012) discovery of CRISPR-Cas9 as gene editing tool;42 (2013) generation of human cerebral organoids from iPSCs;⁴³ (2013) integrating logic and memory with synthetic circuits in living cells;⁴⁴ (2014) complete synthesis of functional eukaryotic chromosome;⁴⁵ (2016) optogenetic control of spinal motorneuron-skeletal muscle contractility unit.

	Cell sources	Individual subsystems	Integrated systems
Examples	ESCs, iPSCs, adult SCs, primary cells, and cell lines	Tissue constructs Neural networks Vascular networks Simple organoid models	Multi-tissue constructs (e.g., neuromuscular units, myotendinous units) Vascularized tissues and organoids Multi-organoid assemblies
Characteristics	Proliferation capacity Self-renewal	Lower level of hierarchical organization Elementary function Immature phenotype	Higher level of hierarchical organization Complex, coordinated function Mature phenotype
Developmental processes involved	Cell differentiation	Pattern formation Self-organization Stochastic decision-making	Formation of interfaces Reciprocal inductive interactions Activity-dependent adaptation
Corresponding <i>in vivo</i> timeline	Very early embryonic	Early to late embryonic	Late embryonic to postnatal

TABLE I. M-CELS hierarchy ranging from cells (parts and devices) through discrete modular subsystems integrated into complete multi-tissue, multi-organ systems.

are essential. This hierarchical organization of development is ubiquitous in nature.

Here, we begin our discussion of M-CELS' design by considering the principles related to autonomous and engineered development of individual modules at the first level of the hierarchical framework. We discuss the developmental tenets essential for driving desired differentiation and organization of M-CELS.

A. Pattern formation incorporating self-organization

In multicellular organisms, groups of cells working together can adopt different fates and organize themselves in space and time to create complex structures. To make these sophisticated heterogeneous shapes, in early animal development, cell fate is decided based on spatiotemporal patterning, which organisms establish and maintain on their own with few, if any, external guiding cues. If we aim to leverage self-organizing spatial patterning in M-CELS, a starting point is to generalize mechanisms and design principles gleaned from natural developmental systems. To illustrate the success and ongoing progress in this area, we first discuss some fundamental tools of patterning in biological development and then examine the recent advances toward synthetically engineering spatial self-organization.

Hierarchically, the cell-cell signaling involved with spatial patterning during development may involve autocrine, paracrine, juxtacrine, intracrine, or endocrine factors. Long-range cell-cell communication is a common tool used to establish spatial patterning in development. A cell can measure and respond to the concentration and duration of a signal that it receives, thereby determining the distance to the source of the signal [Fig. 2(a)]. Using this mechanism, cells acquire their fate based on their relative position in the embryo. Synthetic molecular signal gradients have been demonstrated in mammalian cells using a reconstituted Hedgehog pathway.⁵¹

Long-range cell-cell communication can also be the basis for complex self-organizing patterns. First proposed by Alan Turing in 1952, Turing patterns are a set of patterns of stripes or spots that can arise from a homogeneous initial condition using only differentially diffusing signaling molecules that activate or repress each other. An example of utilizing Turing-based pattern formation in early stage M-CELS was demonstrated in bacterial populations.⁵² This effort required engineering a slowly diffusing activator and a fast-diffusing inhibitor. A demonstration of long-range pattern formation via Turing-like mechanisms in synthetic mammalian systems is still needed.⁵³ The spatial distribution or release of soluble diffusible factors play a key role in generating the radial or longitudinal asymmetric patterns of cellular identities during early embryonic patterning.

Similarly, juxtacrine-like patterning systems have recently been implemented in synthetic living systems. In particular, this effort has included the generation of a synthetic Notch (synNotch) system consisting of the core regulatory domain of Notch, a signaling receptor prolific in animal tissues, linked to a chimeric recognition domain and a chimeric intracellular transcription domain that drive downstream transcriptional outputs.⁵⁴ This tool was used to generate spatial



FIG. 2. Pattern formation and self-organization in M-CELS. (a) Concentration gradients of morphogens can be generated externally or be formed in the presence of a localized source and are transported across the tissue with cell internalization acting as a sink. These morphogen gradients establish spatial patterns of cell types by regulating developmental programs. This strategy for patterning M-CELS usually results in the definition of a relatively small number of distinct regions. Inductive signals provide the patterning guide for the cells to autonomously self-organize into the relevant functional units. (b) Multiple SynNotch receptors can generate multi-layered self-organizing epithelial patterns. The epithelial layer of sender cells and a clonal population of receiver cells are co-cultivated at 1:50 ratio for 10 (start), 34 (day 1), and 58 h (day 2). Reprinted with permission from Morsut *et al.*, Cell 164(4), 780–791 (2016). Copyright 2016 Elsevier. (c) The stochastic symmetry-breaking process occurs within the homogeneous cell mass through sto-chastic activation of genes leading to differentiation and cell specialization yielding cell subpopulations within the larger cell mass. (d) Specific transcription and growth factors can control lineage-specific differentiation from stem cells in M-CELS.

patterns by driving cell–cell sorting through expression of specific cadherin molecules as the downstream signal of synNotch activation [Fig. 2(b)].^{55,56} Cell sorting through differential adhesion has long been known to drive pattern formation in cell aggregates⁵⁷ and is capable of creating an array of different types of structures.^{58,59} Toda *et al.* showed that the combination of synthetic juxtacrine signaling and morphological changes to cell aggregates enabled emergent pattern formation through sequential rounds of patterning of multicellular aggregates, demonstrated by the creation of both two-layer and three-layer structures. It is expected that further engineering of synthetic chimeric transmembrane signaling receptors will enable the fabrication of cell systems with novel morphologies and functions and will also

provide a model for testing the dynamics of intercellular interactions in M-CELS.

Geometry also plays a critical role in the outcome of cell fate determination. The control of geometric confinement and size of the system was shown to determine the number of germ layers specified in embryonic stem cell (ESC) cultures.⁶⁰ Going beyond the 2D flat matrix of a Petri dish, 3D printing and embedding cells within soft gel matrices will permit testing of the effects of iatrogenic geometric positioning on the early development of M-CELS.

Finally, patterning of cell populations can rely on purely synthetic solutions as employed by optogenetics or magnetogenetics. Complex patterning of M-CELS will likely need to employ multiple strategies in

an iterative fashion, in effect mimicking the developmental programs of natural systems.

B. Engineering stochastic decision-making

Natural systems have evolved diverse genetic and mechanical mechanisms to process stochasticity during growth and development. Fluctuations in gene expression (i.e., "noise") are processed by the cell's gene regulatory networks.^{61,62} Gene expression noise can be experimentally minimized,63-65 exploited,66-68 or ignored. Biomolecular feedback loops are involved in regulating stochasticity in cell growth and patterning.⁶⁹ Mechanical forces, mechanosensing, and mechanotransduction pathways can feedback into the cell and initiate gene expression programs that specify cell fate.70,71 M-CELS present an opportunity for interrogating and processing stochasticity in structures composed of multiple cell types and diverse cell-to-cell interactions, which may require a combination of both intra- and inter-cellular approaches. Engineered systems may yield hybrid components that minimize, enhance, or exploit gene expression noise in one parameter regime while ignoring noise in another, such as autoregulation.^{61,72} In M-CELS, stochastic design can orchestrate cell fate determination and cell diversification [Fig. 2(c)].

1. Stochasticity and noise

The ability to engineer stochastic cellular decision making during early development benefits the de novo design and synthesis of M-CELS. Despite the highly regulated network for embryonic stem cells' (ESCs) differentiation control by Nanog, Oct4, and Sox2, the expression of Nanog has large cell-to-cell variability that can be tuned through transcriptional noise.73-75 Stochastic mechanisms produce diverse cell fates across a cell population or tissue. The use or amplification of stochastic genetic decision making achieves the symmetrybreaking differentiation from an initial homogeneous cell mass into distinct cell populations. Approaches intended to emulate stochastic decision making have led to symmetry breaking from an initially homogeneous cell population, resulting in the generation of distinct cell populations.⁷⁶ This symmetry breaking is needed to produce multiple cell types from one cell type without the use of external cues. Stochastic mechanisms can either generate random patterns of cells with different fates or can act as intermediate steps directed by developmental programs to yield reproducible fates.⁷⁷ In both cases, stochasticity plays a role to diversify cell fates.

Here, we present engineering of stochasticity, cell fate determination, and heterogeneity as a design principle for M-CELS.⁷⁸

One such design principle is the dampening of stochastic fluctuations or noise inherent to biological processes to create more homogeneous and reliable systems. This is needed when cell type decisions require unambiguous outcomes (e.g., liver cell vs neuron) with defined ranges of behavior (e.g., enzymatic activities in the liver vs action potentials). Similar to regulation found among ESCs' transcription factors (TFs) and feedback reported to stabilize viral fate determination,⁷⁹ at least 40% of *E. coli* TFs are enriched with negative autoregulation.⁸⁰ These are likely to increase stability and reduce noise magnitude.^{63,79} Negative feedback also acts as a high pass filter, passing fluctuations with higher frequencies, which can more easily be filtered out by downstream genes in genetic circuit cascades.^{61,81,82} The fundamentals of noise or fluctuations in gene expression have been studied for their sources and consequences.^{68,83,84} Gene expression noise is conventionally quantified by magnitude, frequency, and expression levels^{72,85} and can be measured in mRNA and protein populations. The measurement of additional sources of stochasticity, such as cell growth rates and mechanical stresses, is typically performed using quantitative single-cell microscopy imaging.^{86–88}

2. Approaches to modulate biological noise

Noise is a design element that can be actively controlled.⁷⁸ To date, two main approaches have been implemented for controlling gene expression noise and heterogeneity-synthetic biology and pharmaceutical drug treatments. Synthetic gene circuits have been applied as a modular design approach to synthesize artificial tissue and control heterogeneity.⁸⁹ Computational and theoretical analyses of synthetic gene modules for generating population level diversity created a system with increased robustness to uncertain environments. Toda et al. engineered mammalian cells to self-organize into multilayered spheroids using synthetic cell-cell signaling.⁵⁵ Specifically, a lateral inhibition gene circuitry provided cross-repression between adhering cells, resulting in cell fate bifurcation. The authors also demonstrate reliable programming of asymmetric or polarized structures by directing subsets of cells to express variable amount of cell adhesion molecules. In another study, engineered bistable gene networks were implemented in yeast to control stochastic and permanent cell fate determination.9 By combining synthetic gene regulatory networks with natural regulatory networks, the authors achieve stochastic and irreversible binary cell fate determination. Finally, an overexpressed signaling pulse of GATA6 in polyclonal infected human induced pluripotent stem cells (hiPSCs) provided diversification of differentiated cell types into those found in the three germ layers of the embryo (that is, mesoderm, endoderm, and ectoderm fates).⁹¹ Additional efforts have synthetically designed noise in gene expression by engineering *cis* regulation at the promoter level^{92,93} or by the optogenetic control of gene expression level and noise of negative feedback gene circuits.

Exogenous drug treatments have been applied to modulate gene expression noise and bias cellular decision making in cell populations without the need to design and integrate synthetic gene circuits. Noise enhancers and suppressors of gene expression from the HIV long terminal repeat (LTR) promoter were screened for, and combinations of noise modulating treatments were shown to provide synergistic reactivation of latent HIV with enhanced control of viral decision making.⁹⁵ Inhibitors of the BAF nucleosome remodeling complex were demonstrated to fine tune noise in the gene expression of the HIV LTR promoter.⁹⁶ Cell-cycle arrest was used to redistribute the cell-cycle distribution of T-cells latently infected with HIV in order to control a cell-cycle and noise dependent viral fate-decision.⁹⁷

Engineering noise in multicellular systems must also account for sources of heterogeneity and morphogenesis that may not directly result from stochastic gene expression. One such example is how morphological and physical constraints maintain homeostasis in the mouse ear epidermis by filtering stochasticity from a layer of randomly differentiating basal cells.⁹⁸ Asymmetric growth has been identified to maintain homeostasis and carefully balance self-renewal and differentiation in stem and progenitor cells.⁹⁹ Another example is stochastic development in the early embryo, which includes coordination of stem cell sorting and positioning along with stochastic gene expression.^{100,101} Early segregation of the mouse blastocyst into epiblast ectoderm and primitive endoderm is sequential with noisy assignment of cell fates occurring prior to cell sorting.^{102,103} Cell sorting has been implicated as a mechanism for patterning and morphogenesis.¹⁰⁰

Collectively, while continuing to advance the fundamentals of stochastic design and control, our ability to tailor phenotypic diversity by guiding stochastic decision making in M-CELS is achievable. Engineering decision making in M-CELS will require navigating the influence of elaborate intracellular gene regulatory networks, selforganization processes, intercellular lateral forces, and biochemical communication and environmental signaling.

C. Temporal aspects of differentiation

In attempting to recapitulate in vivo development, in vitro cultures progress through a temporally defined series of transition states during differentiation [Fig. 2(d)]. Certain early states are universal across cell types through this process, while some diverge in a hierarchical march as they incrementally mature toward lineage specific checkpoints. Evidence for these states can be measured through gene,¹⁰⁴ protein,^{105,106} and transcription factor^{107,108} expression. Traditionally, these changes have been recorded and observed at the population level. The advent of recent cell specific techniques, such as single-cell RNA sequencing,109 as well as accompanying increase in computational power and dimensionality reduction^{110,111} of large data sets, has provided greater resolution into the cell specific dynamics accompanying this temporal window of differentiation. Through these techniques and analysis of multiple stem cell lineages, a universal understanding of this hierarchical march from early to middle to late differentiation has been identified.

While this hierarchical progression seems to be unidirectional *in vivo*, the function of transcription factors as master regulators of these processes has allowed manipulation of these naturally occurring sequences. In this regard, somatic cell phenotypes can be experimentally reverted to more undifferentiated states¹¹² as well as across distinct lineages. For example, the emergence of common myeloid progenitor cells from the multipotent progenitor stage depends on a dynamic interplay of UP.1 and GATA.1, which persists to determine lymphoid and myeloid lineage specific differentiation further along myeloid cells terminal differentiation.¹¹³

Across development, the same gene may have different functions in different organs. Thus, developmental genes greatly reduce the genetic informational load to direct different stages of development. Most of the regulatory factors guiding development can be grouped into two classes: (1) transcription factors and (2) signaling molecules. Broadly, however, following environmental cue or stimulus, transcription factors and accompanying gene units sets the wheels in motion to route undifferentiated cells, whether they be totipotent, pluripotent, multipotent, or unipotent, down a specific lineage path. For example, upregulation of the transcription factor MyoD alone is enough to shift the ultimate cell fate toward myoblast fate because it serves as a master regulator for multiple downstream genes of this lineage.¹¹⁴

While general overarching principles for early initial differentiation may exist, such as the need for a stimulus to initiate an active progressive drive toward more and more specialized, honed, and specific cell types, more nuanced and specific alterations also play a major role in determining ultimate cell fate and ultimately M-CELS' health, stability, and function. The level of influence of these activities, as well as the specific genes, proteins, and transcription factors involved, are dependent on cell type starting state of differentiation.

1. Replication and maintenance of pluripotent state

The biophysical microenvironment surrounding the stem cell plays a key role in regulating its pluripotency and cell fate determination into specific cell lineages.^{115–118} Intrinsically, at the genetic level, the pluripotent stem cell state is maintained from the expression of a core of transcription factors, such as Oct4, Sox2, and Nanog, along with the specific methylation state of DNA histones epigenetically. The pluripotent state can be terminated by signals extrinsic to the cells, such as soluble growth factors or the extracellular matrix (ECM).¹¹⁸ Growth factors form a major group of extrinsic signals to facilitate or suppress differentiation. While some growth factors promote differentiation, others will maintain the pluripotent state, including transforming growth factor- β (TGF β) superfamily,^{119–121} fibroblast growth factors (FGF),¹²² and insulin-like growth factor.¹²³ It is the balance of intrinsic and extrinsic signals that determine the degree and state of pluripotency-vs-differentiation.

D. Top-down spatial organization: Microenvironment, compartmentalization

Tissues have well developed spatial organization of cells and ECM that arises from the interaction of morphogen gradients, mechanical cues, and geometry.¹²⁴ Organoids recapitulate some of the spatial organization, but to date, the approach is imperfect. The cells undergo symmetry breaking events and self-organization, and the resulting organoids do exhibit a number of features resembling organs. Without extrinsic cues, however, the overall organoid structure is disorganized, and the structures are heterogeneous in size and shape. Specifically, the overall process can be improved by better controlled conditions that produce consistent initial constraints, control of geometry and morphogen gradients to standardize symmetry breaking events, and dynamic boundary conditions that evolve as the organoids develop.^{125,126} These approaches can be extended to produce functioning M-CELS that model tissues.

Sorting early stage spheroids based on specific criteria, such as size and shape, does improve organoid yield, but other factors need to be considered to further improve yield.¹²⁷ While culture of kidney organoids in a 96- or 384-well format produced more uniform structures, the wide range of kidney and non-kidney cell types present highlights the need for more precise control of the conditions regulating organoid formation.¹²⁸ Other approaches include the presentation of controlled mechanical cues such as the stiffness of the substrate to generate kidney organoids with higher differentiation features highlighting on the possibility to emulate native tissue mechanical properties as a new approach to better control organoid differentiation and function.¹²⁶ A microfluidic device can supply morphogen gradients to control pattern formation during germ line development.¹²⁹ This system provides a very useful tool to examine the role of different features, such as temporal variation of the gradient and cell density, to regulate differentiation.¹²⁹ Organoid differentiation can be improved by culture in bioreactors that provide a well-mixed fluid environment that improves oxygen and nutrient transport.^{130,13}

Mechanical properties of the matrix influence the differentiation of organoids in a complex manner.¹³² The different stages of intestinal organoid development require different substrate biomechanical properties. Incorporating a degradable polyethylene glycol (PEG)¹³² created a mechanically dynamic gel that softened over time more closely matching the biomechanical environment needed for growth and differentiation. Other ways to control the environment include the use of hydrogels with light-activated patterns, chemically programmed patterns, or intrinsic control of cellular stochastic processes through synthetic gene circuits as described in Sec. II B.^{125,133} A key challenge to address is how to create dynamic systems that can adapt as the organoid develops that are yet simple enough for high throughput applications.

For various types of M-CELS, spatial organization can be obtained by creating separate microenvironments for different cell types and/or combining with gradients of key signaling molecules to guide interactions. In the fabrication of engineered blood vessels, spatial organization is obtained by extruding endothelium and smooth muscle cells in one annular region and alginate in the outer annular region.¹³⁴ Matrigel attaches to the inner surface of the alginate layer during cross-linking leading to the selective movement of the smooth muscle cells to the alginate interface and endothelium toward the lumen, which forms within a day of fabrication. This vessel structure exhibits a key vessel function, such as contractility and barrier function. While the mechanism has not been established, the authors argue that the vessel structure arises from differential adhesion of the smooth muscle cells to Matrigel and the resulting traction forces.

M-CELS' models of the intestines involve the creation of spatial, chemical, and mechanical gradients in multiple directions. Microphysiological system models have considered the intestinal barrier as separating the acidic intestinal lumen containing microflora from the vascular system. These systems consist of intestinal epithelium in contact with the intestinal lumen contents, extracellular matrix, porous synthetic membrane, and endothelial layer in contact with the culture media mimicking the blood and plasma.¹³⁵ Several key features represent an advance over the simpler Transwell system: flow is introduced into the vascular and intestinal luminal compartment and the membrane is deformable, enabling simulation of peristaltic motion imparting contractile forces on the epithelium and endothelium. Consequently, the epithelial layer begins to develop folds and microvillous-like structures similar to those observed in the intestine. Interestingly, such structures appear with both primary human epithelium and the Caco-2 human colorectal epithelial cell line. Interestingly, these cells formed basal crypts, exhibited microvilli formation, and differentiated into goblet cells, Paneth cells, and enteroendocrine cells found in the intestinal layer.¹³⁵ These systems have been adapted to incorporate cells from human intestinal organoids allowing studies of individual differences in intestinal transport.¹³⁶ Clearly, the presence of mechanical stimuli is critical for differentiation of intestinal cells.

To produce more ordered intestinal structures, Wang *et al.*¹³⁷ used a polydimethylsiloxane (PDMS) mold to create villus structure of the small intestine using micromolded cross-linked collagen on a Transwell membrane. The cross-linked collagen elastic modulus was low enough (\sim 10 kPa) to enable epithelial cells to form a uniform layer without contracting. To produce a gradient in proliferating and differentiated cells found in intestinal crypts, the authors created

gradients of the morphogens Wnt-3A, R-spondin 3, and noggin. Adding the gamma-secretase inhibitor DAPT together with these gradients led to the formation of enterocytes above the highly proliferative cells in the crypt. Collectively, these studies emphasize the importance of matching the substrate mechanical properties with the cell mechanical properties and the importance of morphogen gradients in establishing a variation in differentiation.

III. DEVELOPMENT PART II: INTEGRATION OF MULTI-CELLULAR MODULES

In the hierarchical organization framework of M-CELS' development, the higher level comprises biological systems whose ontogeny includes the integration of constituent modules. The development of the individual modules is largely autonomous and involves the processes of cell differentiation and spatiotemporal organization discussed in Sec. II.

Integrated systems, on the other hand, rely on bidirectional inductive feedback between the constituent modules for the proper development of a system with a specific form and function. Thus, articulating design principles related to the development of integrated M-CELS requires identifying the key constituents of integrated biological systems, investigating the roles of reciprocal inductive cues between constituent modules and developing methods for modulating or mimicking such cues to engineer the *in vitro* development of integrated systems.

A. Development of integrated M-CELS through co-culture

An interesting example of an integrated biological system is provided by the model organism *Drosophila* whose larvae exhibit peristaltic crawling motion coordinated by central pattern generating (CPG) neural circuits. When feedback from sensory organs to the central nervous system during development is inhibited, the CPGs still develop normally, and the larvae generate peristaltic movements, however, the locomotion patterns in these larvae are abnormal.¹³⁸ In this example, the CPG is an individual module that develops autonomously through differentiation of neurons and their organization into a circuit that generates patterned output. Similarly, the muscles and sensory organs are separate individual modules. The entire locomotor system, on the other hand, is an example of an integrated system where interactions between CPGs and sensory organs are critical to attain the desired specific function (forward locomotion).

Naturally, a possible approach for developing integrated M-CELS is to identify the key constituents necessary to attain the desired function and establish appropriate co-culture conditions. A successful implementation of this design principle is illustrated by the bloodbrain barrier (BBB) model of Campisi *et al.*¹³⁹ While a monoculture of endothelial cells is sufficient to establish a microvascular network, these networks do not have the same physiological properties of the BBB. Campisi *et al.* demonstrated that vascular networks developed by co-culturing endothelial cells with pericytes and astrocytes attain the required levels of barrier permeability. In their work, the authors showed that the vascular surface of the membrane contained a confluent layer of brain microvascular endothelium derived from hiPSCs. This layer exhibited tight junctions and restricted permeability with cells expressing specific transporters for glucose. The neural component contains pericytes, astrocytes and, in some cases,

neurons. Pericyte and astrocyte interactions stabilize the endothelium, further reducing permeability and maintaining it for extended periods of time.¹⁴⁰ The presence of flow in the vascular compartment further stabilizes the endothelium.¹³⁷

Hierarchical development is also observed in the musculoskeletal system where the primary structures of bone, tendon, and muscle (i.e., the constituent modules); each develop autonomously through differentiation, migration, and self-organization of the respective progenitor cells. This is followed by the integration of these preliminary structures and formation of interfaces such as bone eminences and myotendinous junctions between the constituent modules (reviewed by Huang).¹⁴¹ Here, the desired function of the system is efficient transmission of contractile forces generated by skeletal muscle to the joints. Engineered muscle-tendon units have been developed by co-culturing tendon fibroblasts and skeletal myoblasts embedded in the ECM in a spatially organized manner. The resulting tissue constructs showed improved tensile and contractile properties.^{142,143}

B. Roles of reciprocal inductive cues

The development of integrated biological systems involves soluble factor-mediated and activity-dependent bidirectional signaling between the constituent modules. These inductive cues, as well as their physiological outcomes, are critical in the design of integrated M-CELS. Appropriate co-culture systems can serve as models to elucidate mechanisms of interaction and eventually help in developing strategies to modulate or mimic them for M-CELS' design. Osaki et al. have recently reported a tissue engineered skeletal muscle-endothelial cell co-culture model where they demonstrated angiogenic sprouting toward muscles and enhanced myogenic differentiation due to synergistic bidirectional signaling mediated by soluble factors secreted by each cell type [Fig. 3(a)].¹⁴⁴ Oh et al. demonstrated that human pluripotent stem cell (hPSC)-derived sympathetic neurons were able to achieve mature phenotype when they were co-cultured with and allowed to innervate ventricular cardiomyocytes, whereas neurons cultured alone remained immature.¹⁴⁵ Similarly, Martin et al. and Aydin et al. observed stronger contractions and improved sarcomere assembly in tissue engineered skeletal muscles that were innervated by motor neurons in co-culture compared to muscles cultured alone.1

The effects of soluble factor-mediated signaling can be recapitulated in M-CELS' development by exogenous supplementation of the factor, thereby allowing phenotypic improvement or maturation without having to rely on co-culture. One approach is to use media conditioned by the target cell type. For instance, skeletal muscle cellconditioned media, when applied to neuronal cultures, including motor neurons, improves neuron viability, enhances neurite extension, and facilitates development of neural networks.¹⁴⁷ Similarly, Schwann cell-conditioned media increases spontaneous neurotransmission in developing motor neuron-muscle co-cultures.¹⁴⁸ If the relevant factors are identified, then it is also possible to supply the pure factor. Agrin, which plays a role in postsynaptic acetylcholine receptor (AChR) clustering in the developing neuromuscular junction (NMJ), can be applied exogenously to induce and/or stabilize AChR clusters.149 Agrin and laminin applied in combination can enhance functional NMJ formation.¹⁵

One must be careful, however, when utilizing exogenous trophic stimulation. Factors that enhance survival and growth may inhibit differentiation and formation of interfaces between cellular modules. Therefore, depending on the relative roles of biochemical cues, exogenous supplementation can impair or facilitate the development of integrated M-CELS. For instance, when *Xenopus* spinal cord neurons were treated with BDNF, GDNF, NT-3, NT-4, Forskolin, IBMX, cAMP, or several combinations of these factors, neurite outgrowth and cell viability were significantly improved. However, this trophic stimulation while keeping cells in the growth state inhibited agrin synthesis in the motor neurons, thus leading to failure of synapse formation when the spinal cord neurons were co-cultured with muscle. Synapse formation was restored when the co-cultures were treated with Schwann cell-conditioned media in addition to the growth factors.¹⁵¹

Another important consideration with regard to exogenous biochemical stimulation is that soluble factors typically elicit dosedependent responses. For example, conditioned extract from muscle tissue enhances neurite outgrowth from motor neurons at lower protein concentrations but inhibits outgrowth at high protein concentrations.¹⁵² When supplementing cultures with commonly used purified growth factors, the appropriate concentrations are typically documented in the relevant literature. However, it may be necessary when using more complex trophic stimulation, such as conditioned media, to perform a dose-response study to ensure the desired developmental outcomes are achieved.

C. Temporal aspects of integration

During the development of integrated biological systems in vivo, the initial formation, refinement, and maintenance of interfaces between the constituent modules often follow a well-defined time course. Throughout the stages of integration, distinct sets of inductive cues act sequentially to orchestrate the development of interfaces. For example, in NMJ development, muscle secreted FGFs, laminins, and collagen IV chains act as cues that induce differentiation of the nerve terminals. FGF7, FGF10, FGF22, and collagen IV 01/2 chains promote pre-synaptic differentiation in the embryonic stage, $\beta 2$ laminins are required for the maturation of embryonic synapses during the early postnatal stage, and collagen IV a3-6 chains are necessary for the maintenance of adult synapses.¹⁵³ A corollary of this insight from developmental biology is that failure to recapitulate the proper sequence of inductive interactions in vitro may prevent the development of integrated M-CELS. For instance, when developing an M-CELS' model of neuromuscular units using stem cell-derived muscles, if the muscle cells secrete FGFs and collagen IV chains but not β 2 laminins, NMJs may initially form but fail to mature since $\beta 2$ laminins are necessary for their maturation. In such situations, endogenous developmental processes may be restored by several methods, including improvement of stem cell differentiation protocols, genetic engineering, or cell sorting.

Exogenous supplementation of soluble factors can also be a useful tool in the development of integrated M-CELS. It is important, however, to consider the time course of integration and ensure that exogenous supplementation is aligned with the developmental timeline. For example, as discussed in Sec. III B, neurotrophic factors (e.g., BDNF, GDNF, NT-3, etc.) that promote neuron survival and neurite outgrowth can inhibit formation of synapses.¹⁵¹ Hence, when developing neuromuscular units *in vitro*, neurotrophic factors can be added during the initial stages of co-culture to improve viability and facilitate



FIG. 3. Development of integrated M-CELS involves interactions among multiple cellular modules. (a) Muscle-endothelium co-culture model reveals the roles of reciprocal inductive cues in angiogenesis and myogenesis. Muscle-secreted factors facilitate angiogenic sprouting, and endothelial cell-secreted factors enhance myogenic maturation. Reprinted with permission from Osaki *et al.*, Biomaterials **156**, 65–76 (2018). Copyright 2018 Elsevier. (b) Role of mechanical cues within the developmental timeline of myotendinous junctions. The formation of force-resistant attachments between myotubes and preliminary tendon structures precedes and is necessary for muscle fiber maturation. Reprinted with permission from Weitkunat *et al.*, Current Biol. **24**(7), 705–716 (2014). Copyright 2014 Elsevier. (c) Microfluidic device designs enable compartmentalization of cellular modules. *In vitro* neuromuscular units are developed by housing neurons and muscles in separate microfluidic channels, connected by ECM hydrogel. Reproduced with permission from Uzel *et al.*, Sci. Adv. **2**(8), e1501429 (2016Copyright 2016 Author(s), licensed under a Creative Commons Attribution 4.0 License. (d) Compartmentalized neuromuscular units are used as an *in vitro* model of ALS by developing co-cultures from patient-derived cells. Reproduced with permission from Osaki *et al.*, Sci. Adv. **4**(10), eaat5847 (2018). Copyright 2018 Author(s), licensed under a Creative Commons Attribution 4.0 License.

axonal outgrowth toward muscles but should be withdrawn once the axons reach the muscle to avoid inhibition of synaptogenesis.

In addition to biochemical signaling, mechanical cues also play key roles in the integration of cellular modules. During the development of myotendinous units *in vivo*, myotubes first extend filopodia toward tendons and initiate attachment. This is followed by an increase in mechanical tension and the subsequent maturation of the muscle fibers and muscle-tendon junctions [Fig. 3(b)]. Interestingly, *in vivo* experiments have shown that the increase in mechanical tension after the initiation of myotube-tendon attachments precedes and is necessary for the maturation of the myotendinous unit. This also means that it is necessary for the initial attachments to be able to sustain mechanical tension.¹⁵⁴ Thus, recapitulating this developmental timeline in M-CELS' models of myotendinous units requires appropriate design of tissue constructs and culture scaffolds to enable force-resistant attachments between myotubes and tendons. A successful design was illustrated by Larkin *et al.* who first developed tissue engineered tendons with highly aligned collagen fibers and tensile properties similar to that of embryonic stage tendons *in vivo*. The authors then integrated these engineered tendons with muscle cells in a culture

scaffold where the resulting tissue bridged between two pins, allowing tension buildup. $^{\rm 143}$

D. Spatial organization: Compartmentalization and microenvironment

In addition to the importance of compartmentalization and microenvironment to cell differentiation (see Sec. II D), these concepts are also vital to the development of integrated M-CELS. Compartmentalized co-culture platform designs may enable successful functional integration of different tissue constructs and/or organoids within a microenvironment that allows for crosstalk and soluble factor delivery. Microfluidic devices are perhaps the most widely used method of spatial organization in an integrated M-CELS design. One example of such an implementation is the neuromuscular unit model of Uzel *et al.*¹⁵⁵ The authors developed a platform where engineered skeletal muscle tissues and stem cell-derived neurospheres are cultured in individual microfluidic channels separated by an intermediate channel. All channels are then seeded with a continuous ECM hydrogel,

which enables the directed growth of axons as well as chemical signaling between neurons and muscles [Fig. 3(c)]. Further developments of this system have enabled *in vitro* modeling and examination of diseases that disrupt or alter the NMJ, such as amyotrophic lateral sclerosis (ALS) [Fig. 3(d)].¹⁵⁶

A key advantage of microfluidic devices is the ability to control and manipulate the fluid medium, which can be leveraged to design compartmentalized M-CELS where different tissue constructs are maintained in separate culture media [Fig. 4(a)].¹⁵⁵ More complex microfluidic platform designs have also enabled the development of organ-on-a-chip models that can capture physiological interactions among multiple organ types and perform metabolomic analysis [Fig. 4(b)].¹⁵⁷ Furthermore, the ability to control spatial organization and fluid flow may also enable the development of vascularized tissue constructs and organoids within microfluidic platforms [Fig. 4(c)], a long-standing challenge in M-CELS (discussed further in Sec. III E).

While microfluidic platforms allow compartmentalization and control of microenvironment within enclosed channels or chambers, certain applications such as biohybrid robots may aim to achieve



FIG. 4. Compartmentalization and microenvironment in M-CELS' designs integrating multiple tissue constructs and organoids. (a) Microfluidic platforms can be designed for the co-culture of different cell and tissue types in controlled microenvironments with separate nutrient delivery to each cell/tissue. Reproduced with permission from Uzel *et al.*, Sci. Adv. **2**(8), e1501429 (2016). Copyright 2016 Author(s), licensed under a Creative Commons Attribution 4.0 License. (b) Microfluidics has also provided a substantial boost for a systems-level understanding of the *in vitro* interaction of diverse organs-on-a-chip, paving the way for the development of a synthetic organism *in toto*. Reprinted with permission from Wang *et al.*, Anal. Chem. **91**(13), 8667–8675 (2019). Copyright 2019 American Chemical Society. (c) Microfluidic platforms may also enable vascularization of organoids, illustrated here by the implementation of a liver organoid embedded within a vascular bed formed by HUVECs and human lung fibroblasts. The immunostaining image shows vessels formed by cells within the liver organoid (magenta alone) integrated with and supported by the vascular network formed by HUVECs (green-magenta colocalization). Unpublished work, images courtesy of Dr. Shun Zhang and Prof. Roger D. Kamm, Massachusetts Institute of Technology, Cambridge, MA. (d) Controlled spatial organization is also possible to achieve on untethered free-standing scaffolds demonstrated here by a biobot design incorporating two separate engineered muscle tissues on a 3D-printed hydrogel scaffold, producing a robotic system capable of multi-directional locomotion. Adapted with permission from Raman *et al.*, Nat. Protoc. **12**(3), 519–533 (2017). Copyright 2017 Springer Nature.

spatial organization of multiple tissues on an open and untethered scaffold, therefore requiring alternative methods. One design approach that has been utilized successfully is modular integration of tissue constructs with 3D-printed hydrogel scaffolds. Raman *et al.* have implemented this approach to design biobots powered by multiple muscle tissues capable of multi-directional locomotion [Fig. 4(d)].¹⁵⁸

E. Nutrient exchange and transport

Vascularization of tissues and M-CELS is a critical step to surpass scaling limitations imposed by simple diffusion. Diffusion alone is insufficient in providing the necessary nutrient exchange and molecular transport necessary to maintain dense, metabolically active cells within large tissues and M-CELS. Vascularization provides the nutrient exchange network to support cell viability in thick tissues and enables endocrine chemical signaling between large cell systems. By engineering vascularization within M-CELS, the chemical and mechanical cues of an *ex vivo* system can be modulated to recapitulate physiological conditions and provide insight into overcoming the scaling challenges involved during *in vitro* organ generation and design of non-natural MCELS.

An essential element in the design of vasculature is the morphology of the resulting microarchitectures following vasculogenesis. Vasculatures can vary through a broad range of scales with human veins and arteries at around 2 cm, venules and arterioles at 100 μ m, and capillaries at around 5–10 μ m in diameter. Venules and arterioles are primarily responsible for the rapid transport of nutrients and waste while capillaries provide the surface area for rapid exchanges of nutrients and cellular molecules. In the context of M-CELS, transitions between transport vessels—capillary beds—transport vessels must have careful considerations in its architecture to consider fluid viscosities, flow rates, friction, and burst pressures to ensure that efficient nutrient exchange occurs while the laminar flow remains throughout the system. While designing the microarchitectures of the vasculatures, the bulk mechanical properties of the construct must also be taken into consideration.

To date, *de novo* vascularization has been engineered either in a cell-based or prefabricated network approach. Cell-based approaches involve endothelial cells or progenitor cells undergoing self-organized vasculogenesis within a tissue construct to generate their own extracellular matrix and undergo lumen formation through chemical signaling.^{159,160} This approach comes with the disadvantage that spontaneous vascularization often requires a significant amount of time and an existing mode for the supply of nutrients. Prefabricated network approaches allow for a predefined architecture within a tissue construct prior to the introduction of cells while also providing a mode for delivery of fresh nutrients.^{161,162} However, the design of such networks is limited by the resolution of existing technologies.

F. Fuzing "living" with "non-living" systems: The biotic-abiotic interface

This past decade has seen fast-paced innovation in the field of biohybrid systems and molecular machines.¹⁶³ These new systems span the entire spectrum from 90% biologic modules with 10% abiotic material to 90% abiotic material with 10% biologic modules. The state of art of current tissue engineering now allows the fabrication of complex living cellular architectures with chemically synthesized ECM

arranged in functional patterns to generate organs-on-a-chip (Fig. 5) or to generate models of tissues such as intestinal tissue, bone marrow, and the blood–brain barrier through 3D bioprinting with living cells.¹⁶⁴

New neuro-electrode hybrids use nanofabrication (CNTs or graphene) to mimic the ECM of the brain and central neural system (CNS) to promote specific neuronal attachment, limit microglial activation, with flexible geometry for shaping to sub-cellular compartments, promote neurogenesis and synaptogenesis, and enable efficient electrophysiological sensing, stimulation, and recording at the neuronal interface.¹⁶⁵ A light-sensitive biohybrid flagella-propelled swimmer was designed, comprising a soft scaffold and live skeletal muscle innervated by motor neurons derived from an optogenetic stem cell neural cluster.¹⁶⁶

A low-power microelectronics controller was embedded in a live jellyfish *Aurelia aurita*, creating a biohybrid robot, and enhancing its propulsive swimming by electrically stimulating muscle contractions at 0.25–1.00 Hz with a square pulse wave (A = 3.7 V, T = 10 ms), thus externally driving its contractions and increasing its peak swimming 2.8-fold.¹⁶⁷ This demonstrated artificial control of animal locomotion addressing actuation, control, and power requirements in soft robotics. A recent paper describes the history and evolution of these devices from biomimicry through biofabrication to biohybrid systems.¹⁶⁸

Biohybrid organoids have been created through the threedimensional assembly of soft, stretchable mesh nanoelectronics throughout the organoid via cell-cell attractive forces from the 2D to 3D tissue configuration during the process of organogenesis.¹⁶⁹ The stretchable mesh nanoelectronics grew and migrated synchronously with the growth of the 2D cell layers into the 3D organoid structure, showing minimal disruption of cell growth and differentiation. The development of this intimate contact between the cells and associated nanoelectronics will allow tissue-wide electrophysiologic measurements at the single-cell level with millisecond spatiotemporal resolution. These single-cell and spatiotemporal measurements are critical for anatomic-physiologic and developmental studies of the heart and brain.

IV. MAINTENANCE, DEGRADATION, REGENERATION, AND ADAPTATION

In addition to designing M-CELS that develop, mature, and ultimately function in a desired manner, it is important to ensure that they can maintain consistent and stable phenotypes and functions over extended time periods. Maintenance of M-CELS is perhaps easy to overlook during design and assembly, but it exists on multiple levels from cellular to tissue and organ-scale, and several contexts such as homeostatic, regenerative, and adaptive. Homeostatic maintenance involves ensuring M-CELS exhibit a desired "baseline" phenotype that does not significantly change or degrade over time. Regenerative maintenance consists of repair mechanisms (e.g., inclusion of a stem cell niche that can be stimulated to replenish injured or degraded cells/ tissue over time) to give M-CELS the ability to respond to acute injuries and extend the functional lifespan of M-CELS. Finally, adaptation refers to designing M-CELS that can adapt to changes in environmental conditions and/or significant modulations to produce persistent alterations in structure and/or function (e.g., muscle hypertrophy in response to exercise). Incorporation of all of these design principles is vital to successful long-term M-CELS maintenance and function.



FIG. 5. Homeostatic maintenance and engineered repair mechanisms in M-CELS. (a) The cellular microenvironment must enable M-CELS to maintain cell identity, viability, function, and the capacity to respond to various signals that may facilitate their integration into larger systems (nutrients, growth factors, cell–cell communications, and interactions). (b) Homeostatic maintenance of the microenvironment to ensure stability of the M-CELS' phenotype with closed-feedback loops and bioreactor-like control systems. (c) M-CELS must be engineered from stem cells and immune cells to enable self-repair and autonomous response to injury. Adapted with permission from Raman *et al.*, Adv. Healthcare Mater. **6**(12), 1700030 (2017). Copyright 2017 John Wiley and Sons.

A. Maintenance, degradation, regeneration

On a cellular level, homeostatic M-CELS' maintenance can be thought of as maintaining the appropriate cellular microenvironment, including (but not limited to) temperature, pH, nutrient exchange, ECM deposition/remodeling, and material considerations (i.e., scaffold/device interactions) (see Ref. 170 for a good overview of design principles for 3D cultures and microenvironment) [Fig. 5(a)]. Proper temperature and pH control are relatively simple to achieve with incubators (including incubated microscopes and other equipment) and physiological buffers and can help in ensuring that cells are

maintained at ideal and reproducible conditions.¹⁷¹ Another solution to maintain steady pH is to use continuous perfusion to replace culture media constantly. These systems, while potentially be difficult to design and implement, have additional benefits of ensuring that cells do not deplete nutrients faster than they can be replenished via intermittent media changes. Perfusion systems have traditionally been used for large-scale cell culture and manufacturing but have recently seen use in microfluidic systems to overcome concerns of impaired cell growth and function due to inadequate nutrient availability.^{172–174} These principles should be applied to M-CELS' design to ensure continuous nutrient availability and ideal pH.

In addition to common nutrients, certain cell types require various growth factors and specific signaling to maintain phenotype-specific genetic programs and homeostasis. Many of these signals are mediated by interactions with, elasticity of, and remodeling of ECM (see Refs. 175 and 176 for summaries of mechanotransduction and matrix remodeling in 3D), which can have pronounced effects on cell phenotype and fate determination.^{177,178} Matrix deposition and remodeling should be considered during M-CELS' design, as artificial matrices or additional cell types (e.g., fibroblasts) may be needed to ensure cells that have the appropriate type(s) of ECM and do not degrade it too quickly (see Ref. 179 for a good review of ECM interactions and challenges with the incorporation of natural and synthetic ECMs). Recently, cysteine cathepsins have been demonstrated to have strong fibrinolytic activity, contributing to the degradation of M-CELS or other engineered tissues containing fibrin.¹⁸⁰⁻¹⁸² As cathepsins are highly secreted by endothelial cells, this serves as an important consideration especially in vascularized M-CELS or may serve as a target for manipulation of degradation rates; however, dosage optimization is crucial to avoid completely restricting the ability of cells to remodel the surrounding matrix, as these interactions remain important for maintenance of homeostasis and cellular function.

Finally, material characteristics and potential biological interactions must be carefully considered when designing M-CELS. For example, polydimethylsiloxane (PDMS) has been widely used in microfluidic devices due to ease of fabrication, optical properties, flexibility, and permeability; however, it was later discovered that PDMS leaches oligomers into media and absorbs lipophilic factors from media, which can have pronounced effects on cell signaling and cellular phenotypes.¹⁸³ While this is not to say PDMS cannot be used for microfluidic fabrication, these interactions are a prime example of the types of considerations that must be made when choosing materials to ensure that materials are not affecting cellular phenotypes or media composition/nutrient availability.

In the case of acute injury or degeneration, regenerative capacity may be needed, such as harboring stem and progenitor cells in engineered stem cell niches. Stem and progenitor cells are capable of expanding and differentiating into various cell/tissue types; therefore, they could be incorporated into M-CELS as a source for regeneration and repair when necessary. While the conditions for maintaining stem cells in undifferentiated states can vary depending on specific cell types and states, many of the same considerations with the microenvironment apply as above. More specifically, the specific niche environments and ways to engineer these niches have been described and are actively being improved,^{184–186} providing an achievable way to incorporate these cells into a larger M-CELS construct.

On the tissue or organ level, it is important to design M-CELS in a way to protect them from physical injury or to implement solutions to repair them in the event of a physical injury. These solutions may include stem/progenitor cells, as mentioned above, inclusion of immune cells (as macrophages, among others), or a combination of these approaches. Juhas et al.¹⁸⁷ demonstrated the use of stem/progenitor cells for tissue repair using engineered muscle derived from primary cells. The authors first differentiated myogenic satellite cells to create engineered muscle tissue capable of contraction and force generation, as well as harboring a small population of undifferentiated satellite cells, similar to natural muscle tissue. After subjecting this engineered tissue to a cardiotoxin insult, the authors observed satellite cell proliferation and eventual differentiation and replacement of the injured tissue over the following 10 days. Raman *et al.*¹⁸⁸ demonstrated that similar satellite cell-driven repair of lacerative tears could occur in engineered muscle derived from a cell line [Fig. 5(c)]. Moreover, the pace of repair could be increased through local controlled release of growth factors at the site of damage and exercise of the muscle tissue, resulting in tissue contractility recovery within 2 days. These strategies highlight the potential for stem/progenitor cells to be implemented into M-CELS and replace or regenerate tissue in response to potential damage that may occur over time.

Similar strategies could be applied to other tissues, such as nerve tissue. There have been many studies on peripheral nerve regeneration (reviewed by Ref. 189) utilizing neural (and other) stem/progenitor cells, various scaffolds (both natural and synthetic), and growth factors to improve nerve regeneration after injury. While there are countless combinations of these parameters with varying limitations, many of them have shown promise in improving nerve repair and have significant potential for neuronal repair in M-CELS. Taking these examples together, one can envision how M-CELS incorporating neural and muscle tissues for actuation could be autonomously repaired with the proper stem cell niches and design. A final consideration regarding repair should be considered when choosing materials for M-CELS' design and should a scenario arise where the scaffolds and noncellular materials comprising M-CELS that need to be repaired or replaced.

Taken together, all of the above aspects require consideration, and potentially engineered control, when designing M-CELS to assure long-term survival and functionality. Maintaining the cell and tissue microenvironment allows for cellular phenotype stability and resistance to degradation, while harboring stem/progenitor cells provide the capability for M-CELS self-repair and regeneration after degradation or injury.

B. Adaptation

Once M-CELS are maintained for extended periods of time, one of their important properties is that they can adapt to changes in environmental conditions, reflecting the response of organisms and tissues. Such responses represent the functional behavior of the various cell types. For example, an increased force on muscle cells produces hypertrophy to enable the cells to sustain increased loads. These changes can be beneficial, such as in cardiac and skeletal muscle hypertrophy in response to physical stimuli (e.g., exercise), neuronal connections due to learning and conditioning, regeneration and repair after injury, or pathological responses such as heart failure or muscle atrophy through denervation, disuse, or disease.

The adaptive response of skeletal muscle M-CELS to electrical, optical, or mechanical stimulation has been studied extensively.¹⁵ These stimuli model the effect of various loading conditions that simulate exercise or conditioning. Electrical stimulation models neural stimulation, producing depolarization of the muscle and initiating contraction. Optical stimulation acts downstream to membrane depolarization, opening calcium channels. Mechanical stimulation models contraction and extension, subjecting muscles to strain and forces similar to those encountered in vivo. In muscle and other cells/tissues (e.g., endothelial cells), cells respond substantially to various biomechanical forces, including relatively minor/short-term responses (e.g., mechanically gated channel opening) and major/long-term responses (e.g., altered gene expression). Combinations of different modes of stimulations can synergistically produce greater increases in forces than observed with one mode of stimulation alone.¹⁹¹ Removal of load stimuli can result in losses of force production and atrophy of engineered muscle, as with native tissue.¹⁹² M-CELS designed with adaptive responses such as these have functional advantages and are better-suited for long-term survival and ability to perform a wide array of tasks/functions.

C. Simulating neural activity: The long evolutionary road to cognition, learning, and memory

Attempts to incorporate the neural functions of learning, memory, and cognition have been quite challenging and the M-CELS' cognitive capacity is in its early infancy. Synthetic biologic cognition and memory is still a distant goal, but progress has been made in organoids and biohybrid tissues. The state of art of brain organoid technology has been recently reviewed.¹⁹³ The stem cell-derived self-organized 3D neural aggregates comprising cerebral organoids¹⁹⁴ have been used as downscaled physiologically relevant in vitro models of the human brain¹⁹⁵ for studies of cognition and neurologic disease.¹⁵ Optogenetic stimulation during neural differentiation can result in permanent changes that extended to the genetic expression of neurons as demonstrated by RNA sequencing.¹⁹⁸ While cerebral organoids have deficiencies,¹⁹⁹ such novel or unique neural functions may be useful and applicable for the practical purposes for which M-CELS are designed or built. Recent studies on engram neuronal cells using optogenetic cell labeling and c-fos-promoter transgenic mice are building a foundation for later engineering synthetic neuronal assemblies with long-term memory, a critical attribute for cognition and learning in M-CELS.²

Biohybrid neural tissues²⁰¹ facilitate two-way transfer of information between neural tissues and external devices. These devices demonstrate integration with distinct regions of the brain with and multiplexed neural recording.²⁰² For example, biofabrication of neural tissue, which forms a three-dimensional neural tissue mimic helps with the assessment of novel neuro-electrodes for the biotic-abiotic interface, and the investigation of neurotransmission and biocompatibility.²⁰³ Neuromorphic neural interfaces now focus on coupling of biohybrid devices with biological tissue.²⁰⁴ Neurobiohybrid swimming machines have been built, which are driven by on-board neuromuscular units and an optogenetic stem cell-derived neural cluster containing motor neurons.¹⁶⁶

The first biohybrid neural synapses have been implemented through the coupling of an organic neuromorphic device with dopaminergic neurons.²⁰⁵ This biohybrid synapse is a leap forward toward the harmonious fusion of biologic neural networks with artificial neuromorphic systems. An earlier biohybrid coupling of neuromorphic and biologic neural networks demonstrated the feasibility of coupling the two networks and implementing control circuits capable of modifying the biohybrid synapse between the neuromorphic and biologic networks.²⁰⁶

D. Noninvasive continuous monitoring of M-CELS

While insightful maintenance of M-CELS is essential for longevity of intended form and function, noninvasive monitoring systems are the crux, which inform maintenance decisions. Without engrained systems for consistent feedback and monitoring of macro- and microenvironments, M-CELS become a black box of unknown perturbations and activity. Therefore, an essential, yet often overlooked principle for designing M-CELS, is inclusion of noninvasive monitoring systems. While engineering concerns, such as cellular components,²⁰⁷ scaffolding,²⁰⁸ and microenvironment, are critical to M-CELS' design, monitoring and feedback from these engineered construct help ensure longevity of M-CELS as well as garner information for reproducibility and applicability [Fig. 5(b)]. These systems designed to observe, measure, and assess should capture an M-CELS' dynamic progression through multiple phenotypic states without the risk of mechanical, chemical, or physical perturbation of the system. This level of observation will endow a better understanding of the effects timescales and environments have on the phenotypic decisions of these systems. This enhanced understanding eventually could permit greater precision over the development of these systems, as well as allow the heightened knowledge of similarities and differences between the checkpoints of M-CELS compared to in vivo developmental systems.

Though multiple monitoring systems and readouts currently exist, a majority of these techniques involve perturbation of the system, which can effectively halt or alter the natural phenotypic progression of M-CELS. Special thought and emphasis should be given to techniques and instruments which allow for continuous or long-term noninvasive monitoring of developing systems. Currently, there exists multiple examples of such monitoring systems that could be easily incorporated in the design schema of M-CELS such as sensor systems,²⁰⁹ describing a low-cost sensory system which can measure the optical density of media. These measurements can then be correlated with effective permittivity to ascertain a reasonable noninvasive measure of growth dynamics of a culture. At higher frequencies, this sensory system could be used to monitor changes in the osmolarity capturing minute micro-level changes of M-CELS which would go unnoticed otherwise. All of this information would serve to better understand the microenvironment of the M-CEL system, and specifically, how it behaves on typically overlooked timescales and measures, as well as how it compares to in vivo systems and other developmental systems.

While sensor systems provide a noninvasive glimpse into the micro-environment of M-CELS, various imaging techniques have been the preferred method for monitoring systems on macro-level scales. Recent advances, however, have allowed for an even closer look at M-CELS' environments, interactions, and intracellular responses. One such technique is digital holographic microscopy (DHM).²¹⁰ DHM allows for noninvasive monitoring of cell cycle arrest and multiple other parameters such as cell number, confluence, and phase volume. In their study, Miniotis *et al.*²¹⁰ were able to directly measure changes and arrests associated with G1 and G2/M phases via application of specific

cell cycle inhibitors. Additionally, spatial light interference microscopy (SLIM) provides label-free assessment and quantification of subcellular biological properties and activity, including dry mass and mass transport.²¹¹ Currently, it is necessary with this method to first conduct base-line studies to confirm cell cycle phase with more invasive equipment such as flow cytometry. Following confirmation, however, DHM monitoring could provide cell cycle specific information from imaging, alone. Similar monitoring systems can be coupled in parallel with additional imaging systems targeted at various scales, such as macrolevel ultrasound²¹² and magnetic resonance elastography,²¹³ which maintain non-destructive imaging structures, yet allow for an even more comprehensive level of understanding of engineering tissues through different phases.

Other noninvasive monitoring systems are possible with genetic manipulation. One common example of this is the use of calcium reporters, which allow for monitoring of functionality in cells and engineered tissues. Juhas *et al.*¹⁸⁷ in their study described about using endogenous satellite cells to repair engineered muscle tissue, demonstrated this approach by genetically implementing a fluorescent calcium indicator, GCaMP3, which allowed them to noninvasively detect and continuously monitor, in real time, intracellular calcium levels in implanted engineered muscle tissue. By monitoring intracellular calcium levels, the authors were able to monitor contractile function and vascularization via intravital imaging. GCaMP3 and other calcium indicators have also been used to monitor neural activity and are applicable to a wide range of cell types and functions.²¹⁴

Overall, noninvasive monitoring is vital for ensuring that M-CELS are maintained and functioning properly, leading to improved readouts and control of various M-CELS' parameters. The ability to monitor precise cellular processes and macroscale functions can assist with optimization/improvement and inform future M-CELS' design considerations.

V. FUNCTION

A significant part of M-CELS' design is implementation and readout of specific functions. Given the expansive variety of cell and tissue types, materials, and engineering techniques available when designing M-CELS, potential functions can vary considerably, ranging from organoids and platforms to study development to complex multi-tissue assemblies with controllable outputs. Despite the vast conceivable range of functions for M-CELS, the design and implementation of these high-level functions typically involve a combination of three main subfunctions: sensing, information processing, and functional output [Fig. 6(a)].

Signals from the cellular microenvironment and surrounding cells, as well as intracellular signals, serve as the main sources of cellular inputs. As discussed above and in more detail in Sec. V A, cells sense many aspects of their environment, including temperature, pH, soluble factors, matrix composition/mechanics/topography, neighboring cells, and other forces (i.e., shear flow or compression).²¹⁵ Additionally, cells can be engineered to enhance this existing sensory capacity or add entirely new sensing capabilities. All of these sensory inputs and information are vital to detect microenvironmental and external changes but must be processed effectively to inform output and actuation. Additionally, the ability to sense intra- and extracellular changes provides M-CELS with the important capability for feedback regulation and adaptation.

Information processing is vital to the control of cellular functions and outputs both homeostatic and directed. While cell signaling networks can be highly complex, there have been countless studies devoted to elucidating common signaling pathways and their involvement in specific functions at molecular, cellular, and organismal levels. The processing of these and other signals and sensory inputs takes several forms, including relatively quick processing (e.g., phosphorylation and signal transduction cascades) or slow, long-lasting processing (e.g., genetic regulation), combinations of which are important in producing complex and effective responses. In addition to single-cellular processing, M-CELS can also be designed with multicellular processing in mind, such as neural network incorporation.

Finally, actuation and output design allow M-CELS to perform various tasks or provide targeted readouts. Actuation allows M-CELS to carry out high-level functions and includes many possible modalities and types, such as mechanical actuation, chemical synthesis and secretion, optical output, bioelectrical output, and structural deposition or modification. Combinations of these output mechanisms can be employed to design M-CELS with many unique functions for a wide variety of applications, some examples of which are detailed below.

A. Sensing

1. Extracellular sensing

An immense diversity of mechanisms exists in cells for extracellular sensing. Receptor tyrosine kinase (RTK) cell surface receptors are sensors for many extracellular signals, including growth factors, hormones, trophic factors, and cytokines.^{216,217} In addition, RTKs' cell surface receptors and cytokine receptors are also another important family of receptors that sense cytokines through their extracellular domain.^{218,219} G protein-coupled receptors (GPCRs) are another major family of receptors used for extracellular sensing. This family of integral membrane protein receptors is involved in sensing chemokines, odorants, hormones, pheromones and photons.²²⁰ Among other sensing modalities and mechanisms, integrins are transmembrane receptors that attach to the extracellular matrix and mediate the response of the cell to the force, rigidity, and ligand distribution in the extracellular matrix.²²¹ Ion channel-linked receptors are transmembrane receptors that respond to chemical messengers such as neurotransmitters.²²² Transient Receptor Potential (TRP) channels are also an important family of ion channels, which use ions for signal transduction and are responsive to several stimuli, including heat, toxins, protons, pressure, and osmolarity.²²

In addition to the sensing mechanisms for extracellular signals discussed, Notch receptors are also able to sense nearby cells.²²⁵ Immune response pathways also have specialized receptors to sense and interact with nearby cells and pathogens.^{226–228} Nuclear receptors can also sense small molecules that diffuse through the membrane such as steroids and thyroid hormones,²²⁹ vitamin D,²³⁰ retinol,²³¹ and endocrine-disrupting chemicals.²³² Synthetic biology also enabled the engineering of sensors for extracellular signals. Examples of engineered sensors include synthetic Notch (synNotch) receptor, chimeric antigen receptors (CAR), engineered light-activated ion channels, and various sensors for environmental monitoring, manufacturing, and health-related applications.^{54,233–235}



FIG. 6. M-CELS integrate and process sensory inputs to generate functional outputs. (a) M-CELS' sensing, processing, and actuation. M-CELS can sense specific extracellular and intracellular stimuli through a variety of sensors. The signals can then be processed to recognize specific levels, patterns, and combinations of stimuli, which, in turn, controls a specific set of output responses in the form of complex biological behavior that can be used for practical applications. (b) Intercellular connectivity of cardiomyocytes arranged in a serpentine pattern on a tissue engineered ray enables a global activity pattern in response to local input. Optical stimulation is applied locally at the front of the fins, and the resulting activation signal propagates via gap junctions through the body length. Republished with permission from Park *et al.*, Science **353**(6295), 158 (2016). Copyright 2016 Author(s), licensed under a Creative Commons Attribution (CC-BY-NC-ND) License. (c) Neuronal actuation and control of muscle-powered biohybrid machines: soft robotic swimmer propelled by flagella, composed of a compliant scaffold, and engineered skeletal muscle tissue innervated by motor neurons derived from optogenetic stem cells. Adapted from Aydin *et al.*, Proc. Natl. Acad. Sci. U. S. A. **116**(40), 19841–19847 (2019). Copyright 2019 Author(s), licensed under a Creative Commons Attribution (CC-BY-NC-ND) License.

2. Multi-cellular sensors

Multi-cellular sensing is pervasive in sensory perception and a wide range of mechanisms and specialized multicellular structures exist in living organisms. In olfactory perception, millions of olfactory sensory neurons (OSNs) form a neuroepithelium in vertebrates, which enables the identification of specific odorant molecules using a multi-cellular combinatorial code.²³⁶ In auditory perception, hair cells of the basilar membrane act in a coordinated manner to process the frequencies in sound, which makes the perception of complex sound

possible.²³⁷ Moreover, specialized multicellular structures that comprise hair cells in the inner ear are also essential for the sense of balance and movement in mammals.²³⁸ Proprioception in mammals is achieved through specialized mechanosensory neurons with multicellular diversity in muscles, tendons, and joints.²³⁹ Visual perception is mediated by the complex multicellular response of photoreceptor cells in the retina, which are capable of phototransduction.^{240,241} Moreover, unicellular organisms can also perform multicellular sensing, such as quorum sensing in bacteria, which enables the detection and response to cell density.¹⁶

3. Intracellular sensors

Intracellular sensing is vital for every organism, and an immense array of mechanisms both endogenous and engineered exists. Sensing intracellular nutrient levels, metabolic flux through specific pathways and their respective gene expression, *E. coli* will utilize the pathways, which maximize their growth rate, and this can be predicted by an *in silico* model.^{242,243} Contrary to the premise that gene expression pathways evolved solely to maximize biomass production, deletion mutants in *B. subtilis* grew faster than the wild type.²⁴⁴ Sensing glucose levels may result in cellular remodeling of nucleoli structure.²⁴⁵ Amino acid starvation was observed to increase phosphorylation of nuclear acidic proteins.²⁴⁶

Cellular metabolite sensing of bioenergetics for allocation of resources and activation of relevant gene expression is a homeostatic biological function essential for survival.^{247,248} These metabolite sensing and allocation functions are best studied based on mTOR and AMPK signaling. Switching among available biochemical pathways is a critical function during cell growth, proliferation, cell cycle, and stress response, which is facilitated by cell signaling pathways.^{249–252} Early work on metabolite sensing and transcriptional regulation was carried out in *Escherichia coli*.²⁵³ NAD-sensing was observed in early studies, which revealed that protein synthesis was inhibited by the ADP-ribosylation of elongation factor Ef-2, which is dependent on NAD+.^{254–256} Metabolic intermediates are also sensed through activation of nuclear receptors.²⁵⁷

Aside from nutrients and metabolites, there are many other mechanisms for sensing cellular processes and molecules. Epigenetic regulation involves the observation of histone acetylation as key to activating gene expression.^{258,259} Bioenergetic switching from aerobic to anaerobic metabolism is heralded by oxygen-sensing of hypoxia, which leads to epigenetic acetylation of histones promoting lipid synthesis.²⁶⁰ Multiple studies have shown that bioenergetic sensing of the cellular energy needs vs fuel levels involves AMPK signaling and mTORC1 signaling as master regulators of cell metabolism. Sensing for intracellular messengers, such as small molecules and ions, is ubiquitous.^{261,262} In addition to the nuclear receptors discussed in Sec. VA2, another important intracellular receptor is the inositol trisphosphate receptor (InsP3R), which is activated by the signaling molecule inositol trisphosphate (IP3).263 Specialized pathways and mechanisms also exist for sensing and regulating proteostasis.²⁶⁴ Moreover, pathogen-associated molecular patterns (PAMPs) sensors, damage-associated molecular patterns (DAMPs) sensors, RNA sensors, DNA sensors, and other intracellular sensors for pathogen detection have been observed.2

Synthetic biology has enabled the ability to engineer novel intracellular sensors. The first papers in synthetic biology focused on engineering genetic circuits analogous to electronic circuits enabling a genetic toggle switch¹⁴ and a genetic oscillator.¹⁵ Promoters and repressors (tet TetR, lac LacI) and transcription factors (lambda phage regulators) served as the prototypical sensors of intracellular conditions to activate these genetically engineered circuits.^{26,61,266–268} Genetic design of metabolic pathways manifest the sensing and regulation of bioenergetics and the carbon cycle, which can be observed in the differential regulation of the lac operon in synthetic *E. coli* where a cost-benefit analysis provides data to parameterize a fitness function for cells as a function of lactose.²⁵³ Cell-based biosensors have also been engineered for sensing metabolites, small molecules, mRNAs, miRNAs, and proteins.^{235,269–271}

B. Information processing by M-CELS

Cell function arises in a multitude of ways. Many behaviors exhibit straightforward relations between the stimulus and the response. Alternatively, delayed changes in the response to a stimulus that result from changes in cell state (e.g., growth, development, differentiation, apoptosis, and senescence) are activated by signaling pathways that activate or repress genes that control these states. Many of the ligands that initiate these long-term events converge on a limited number of signaling pathways. Furthermore, the same signaling molecules can induce different cell fates due to the presence or absence of other signaling molecules or the state of the cell.²⁷² Cells achieve these states using chemical signaling and gene expression, which act like logic gates. Signaling pathways and genetic regulatory elements generate logic gates when the concentration of a key molecule changes dramatically with an input such that the regulatory molecule is either above or below a critical level to activate a response.²⁷³ These logic gates can be used to create new cellular functions, providing new approaches to regulate the state and function of M-CELS.

Advances in genetic engineering and synthetic biology have led to engineered cells and receptors, allowing for modification or addition of cellular receptors and information processing systems. A highprofile example of this is chimeric antigen receptor (CAR) T-cells, which are being used to treat cancer. Recently, other complex systems are quickly being developed to give cells improved sensory and processing capabilities, such as synthetic Notch (synNotch).⁵⁴ Orthogonal synNotch constructs have been implemented, creating receptor ANDgates, representing improvements in information processing ability, and getting closer to engineering M-CELS with the capability of performing logical computational processing.²⁷⁴ The modularity and significant customizability of the synNotch system allows for great flexibility in designing specific responses and functions with control of ligands, receptors, and genes that can be combined in various ways to enable complex cellular information processing. Similarly, genetic Boolean programs^{275,276} represent a powerful way to engineer cells that integrate and process complex inputs, which is also useful for creating cells with desired functions in M-CELS.

In order to improve the specificity of chimeric antigen receptor (CAR) T cell therapy,²⁷⁷ created synNotch receptors that acted as AND gates. The extracellular portion of one synNotch domain consisted of the CD19 receptor to enable T cells to recognize CD19 on tumor cells and an intracellular T cell transactivator. On the other synNotch, the CAR mesothelin receptor was inserted on the extracellular portion, which is the same intracellular activation as the CD19 synNotch. The result was a T cell response that was only induced by tumor cells that expressed CD19 and mesothelin and animal studies showed improved survival to tumors expressing CD19.²⁷⁴

The concept of AND gates was also used in nanorobots that selectively targeted cell subtypes.²⁷⁸ The nanorobot consisted of a DNA origami structure held in place with a clasp with a pair of DNA aptamers. The clasp was released and payload delivered only when both antigens were present. Alternatively, the nanorobot could recognize one target molecule if both aptamers targeted the same molecule. Cell adhesion could be regulated by grafting adhesion peptides or molecules to single-stranded DNA (ssDNA) sequences that can bind to

complementary sequences on the substrate.²⁷⁹ Addition of a ssDNA without the adhesion motif was used to induce dissociation so that AND, NOT, and OR gates can be produced. These examples show the potential of exploiting logic elements to regulate a variety of cell functions.

C. Actuation, readout, external communication

1. Mechanical

Mechanical actuation in living organisms serves a vast range of functions from intracellular transport at the subcellular scale to swimming and flying at the organism scale. In animals, muscle tissues are the primary means of generating mechanical output at the multicellular level. Consequently, muscle cells have been used in numerous studies over the past two decades to develop M-CELS capable of mechanical actuation, reviewed in Refs. 48, 190, and 280

Cardiomyocytes are often cultured as adherent cells on a compliant substrate, either as small isolated clusters²⁸¹ or, more commonly, as multicellular confluent cell-sheets,^{282–287} to develop biohybrid machines capable of locomotion or pumping. One of the key features of cardiomyocytes is that they form gap junctions that connect the cytoplasm of neighboring cells, thereby electrically coupling them. In a multicellular sheet of cardiomyocytes, this gap junction-mediated connectivity can be exploited to design complex global actuation patterns triggered by simple local inputs. An example of such a strategy has been demonstrated by Park *et al.* who cultured cardiomyocytes in a serpentine pattern on a compliant substrate. Here, cardiomyocytes at one end of the pattern were stimulated locally and the evoked action potential propagated through the pattern via gap-junction connectivity [Fig. 6(b)]. The resulting undulatory deformation of the compliant substrate was used for locomotion.²⁸⁵

Skeletal muscle-based actuation has also been implemented in M-CELS to enable the design of biohybrid walkers,¹⁹² swimmers,¹⁶ pumps,¹⁶⁹ and miniature robotic arms.²⁸⁸ In contrast to cardiomyocytes, which are single mononucleated cells capable of sarcomeric contraction, contractile skeletal muscle cells are long multinucleated fibers which form via the fusion of myoblasts. Hence, skeletal muscle bioactuators in M-CELS are developed de novo through fusion of myoblasts and the subsequent cytoskeletal maturation of myotubes, often within a reconstituted 3D ECM. The mechanical output of the resulting muscle tissue, therefore, depends on factors that influence this process of development, such as cell source, cell density, ECM type and concentration, culture conditions, and stimulation methods (reviewed in Refs. 289-291). The relationships between these process parameters and the desired mechanical output can be investigated to guide novel actuator designs or improve existing actuators. For example, Pagan-Diaz et al. have shown that by differentiating myoblasts into myotubes on a rigid substrate prior to embedding cells in ECM resulted in a threefold increase in the contractile force output compared to actuators that were created by embedding myoblasts directly in ECM.²

Effective stimulation and control of muscle contractions are also key considerations for bioactuator design. While spontaneous contractility of muscle cells can be utilized to achieve autonomous function,^{281,282} external stimulation is a more common approach since it offers a means to achieve prescribed contraction dynamics. Cardiac and skeletal muscle cells can be stimulated electrically or optogenetically with precise control of stimulation frequency and intensity.

Electrical coupling of cardiomyocytes via gap junctions, as discussed above, enables robust global actuation by local stimulation. Skeletal muscles, however, typically lack this coupling since connexins (gap junction proteins) are downregulated during skeletal muscle development.²⁹³ ⁴ Evoking coordinated global contractions of a multicellular skeletal muscle bioactuator, therefore, requires effective stimulation of all individual muscle fibers, which can be achieved by appropriate design of tissue constructs and stimulation techniques. In the case of optogenetic stimulation, for instance, the limiting factor is the penetration depth of light. Thus, optogenetic muscle tissues with smaller cross-sectional dimensions can be stimulated more effectively compared to bulkier tissues.¹⁹¹ For skeletal muscle-powered M-CELS, neuronal actuation is also possible and has been demonstrated by the development of a biohybrid swimmer driven by on-board tissue engineered neuromuscular units [Fig. 6(c)].¹⁶⁶

Muscle-based actuation, as it has so far been demonstrated in M-CELS, typically involves the integration of muscle cells with synthetic compliant scaffolds. Scaffold design is, therefore, a significant aspect of M-CELS with mechanical output. Several studies have demonstrated biomimetic scaffold designs which mimic jellyfish,²⁸⁴ stingray,²⁸⁵ or sperm²⁸¹ morphologies for locomotion. Borrowing from nature in this manner is an effective way to design bioactuator scaffolds, especially when the mechanics of locomotion of the organism, that is, being imitated, is well-understood and the designer can capitalize on existing mathematical models. Conversely, biomimetic M-CELS may also serve as experimental models to study the mechanics of the organism being imitated and possibly help in elucidating design principles which could then be applied to conventional robotics.²⁹⁵ Furthermore, it is possible to go beyond biomimetics and develop entirely novel systems using algorithmic design processes. This was recently demonstrated by Kriegman et al. who used an evolutionary algorithm to generate, evaluate, and select novel bioactuator designs that achieved desired locomotor performance.29

2. Chemical

M-CELS can be engineered to secrete different biochemical moieties in response to a variety of external stimuli or triggers, and this has a wide array of potential applications in medicine, food production, biosecurity, and beyond. Cellular chemical factories are especially interesting in medicine, because they enable personalizing a therapeutic drug dosing regimen to the needs of individual patients. The current standard of clinical care generally involves administering small molecules via ingestion and biologics via injection and often requires consistent and complex dosing schedules that reduce patient comfort and compliance. For diseases with known biomarkers, a multicellular factory could dynamically sense changing biomarkers in the bloodstream and adaptively adjust the production of a therapeutic biologic in a personalized and closed-loop manner. This type of adaptive response could dramatically advance human health, especially when applied to chronic debilitating diseases with unpredictable progression.

One such chronic disease, diabetes, affects over 400 million people worldwide.²⁹⁷ Type 1 diabetes is currently treated via regular blood glucose monitoring coupled with insulin injections, a process that is both imprecise and uncomfortable for patients. Insulin is produced by β -cells which reside in clusters within the pancreas known as the islets of Langerhans. The rate at which they secrete insulin is regulated by the level of glucose in the bloodstream, and their loss or dysfunction results in diabetes. Understanding how these cells interact in health and diseased states is critical toward developing and testing new therapeutic interventions for diabetes. As a result, a significant body of research has focused on finding ways to grow and monitor clusters of β -cells *in vitro*.

Since multicellular clusters rely on complex 3D cell-cell and cell-matrix interaction, 2D culture models offer an inadequate representation of the *in vivo* mechanical and biochemical environment.²⁹ 2D culture-driven changes in gene expression and phenotype are especially potent in β -cells, which communicate through EphA receptors and EphrinA ligands in vivo, and lose their ability to produce insulin in 2D environments in vitro. A study by Hammond and colleagues generated a 3D multicellular β -cell construct by growing cells near synthetic "neighbor cells," microscale gel beads biochemically modified to present EphA and EphrinA as well as a range of other extracellular matrix components derived from decellularized rat pancreas.²⁵ Culturing the cells and beads together enabled the cells to interact with each other, and with their synthetic neighbors, while having freedom to migrate. Moreover, void spaces between the cells and beads enabled ready diffusion of oxygen and nutrients throughout the multicellular construct. The researchers showed that this biohybrid multicellular manufacturing method promoted cell survival fivefold and helped in maintaining cell function as assessed by insulin secretion in response to glucose stimulation up to 21 days in culture.

While such in vitro culture systems can lend valuable insight into the onset and progression of diabetes, we must eventually find ways to replace lost or dysfunctional cells β -cells clusters in the body to treat this disease. Transplanting β -cells from cadaveric donors is affected by limited supply of donor tissue. Mature β -cells derived from human embryonic stem cells could be a robust and sustainable source for pancreatic cell therapy.³⁰⁰ However, both cadaveric and stem cell-based donor cells elicit an immune response from host bodies and, therefore, must be coupled with lifelong immunosuppressive therapy. One way to circumvent this is to encapsulate implanted pancreatic islets in hydrogels that protect the cells from the host's immune system. In the past, this approach has generated a foreign-body response in the host that resulted in fibrotic encapsulation and loss of function of the implant. Promisingly, Anderson and Langer and colleagues have developed a new chemically modified alginate hydrogel that resists implant fibrosis. Stem cell-derived β -cells embedded in modified alginate gel spheres (1.5 mm in diameter) were transplanted in diabetic immunocompetent mice.³⁰¹ The researchers observed that normoglycemia was restored and maintained up to 174 days post-implantation in these mice with very little immune response.

Multicellular chemical factories are promising not only for diabetes, but also for other diseases that can be combated through the administration of biologics. One such example disease is psoriasis, a chronic condition characterized by red and white bumpy scaly patches that erupt on the skin unpredictably. The chronic and unpredictable nature of this disease makes it a prime target for a treatment modality that produces therapeutic proteins on-demand in response to autonomously detected disease biomarkers. Fussenegger and colleagues created such a modality by engineering cells that could detect the proinflammatory cytokines associated with psoriasis, tumor necrosis factor (TNF) and interleukin 22 (IL22).³⁰² Importantly, since upregulation of either of these cytokines alone is not specific to psoriasis, they incorporated Boolean AND gate logic into their cellular circuit to ensure that both cytokines would need to be upregulated in order for the cellular factory to be activated. When activated, the cells produced interleukin 4 (IL4) and interleukin 10 (IL10). The researchers encapsulated these cells in an alginate-based hydrogel and tested their functionality when implanted intraperitoneally in a mouse model of psoriasis. Their cell therapy restored normal skin morphology and prevented the onset of psoriatic flares in these mice and, furthermore, was able to detect and respond to TNF and IL22 in human blood samples.

These few examples outline the tremendous power and potential of engineered cellular systems with chemical output response for medical applications. Future studies that develop M-CELS' capability of multi-step decision making in response to mixed external stimuli could lend further depth and diversity to this field.¹³³ Applications outside the realm of medicine, such as in producing large quantities of biological molecules for food production, might rely on non-mammalian cells and industrial scale bioreactors.

3. Optical

Several tools have been generated for the rational engineering of optical reporters to provide dynamic readouts of the cellular decisionmaking processes that take place collectively in M-CELS. These include multi-labeling lineage tracing tools such as Brainbow and Brainbow-type variations.^{303–305} Brainbow constructs utilize recombination events to stochastically express specific combinations of fluorescent proteins (FPs) in cells. This provides a powerful tool for lineage studies where multi-spectral labeling of progenitor cells and their progenies define the extent of clonal expansion of cell types. The number of unique cell types that can be labeled uniquely is constrained by the number of fluorophores that can be independently controlled and clearly separated during imaging. New multi-spectral constructs enable independent switch-like control of fluorescent protein expression within a single construct. For example, the Bitbow system encodes information into the ON/OFF expression state of the fluorescent proteins, with the Bitbow system encoding up to 32767 unique color states.^{306,307}

Fluorescence-based reporters require excitation light sources and these can create issues such as autofluorescence, photobleaching, and phototoxicity. Additionally, light illumination could trigger unwanted stimulation of optogenetic activity. Recently, bright luminescent protein reporters called nano-lanterns have been generated that utilize bioluminescent resonance energy transfer (BRET).³⁰⁸ The energy transfer occurs from enhanced *Renilla* luciferase to fluorescence protein. The resulting cyan and orange nano-lanterns are about 20 times brighter than wild type *Renilla* luciferase. Nano-lanterns enable multicolor live imaging of intracellular structures and could be used in conjunction with optogenetic tools to create a palette of optical input and output communication channels between M-CELS and operators.

Multi-spectral optical reporters could be extended to provide optical readouts of more than cell lineage by utilizing genetic switches that are regulated by additional inputs. CaSSA is a platform to create genetic switches using CRISPR/Cas9 (Ca) and DNA repair mechanisms of single-strand annealing (SSA). Multiple gRNAs are expressed in specific patterns to create cell-type manipulations based on the

expression of cell-specific genes.³⁰⁹ Uniquely addressable switch-like control of fluorophore activity to create multi-spectral digital readouts could facilitate closed control loops that provide optical reporter-based feedback during the dynamic operation of M-CELS. Developing optical-mediated control loops for M-CELS will require the tuning of stimulation (inputs) and outputs (imaging of the optical-based reporters).

4. Bioelectric

Bioelectrical output is another form of output that cells can use for actuation or to report information about their state or the state of their environment. M-CELS incorporating neural networks or organoids could, in theory, use these neural circuits to provide information on their surroundings, especially given recent advances in electrophysiological recording, including microelectrode array (MEA) technology and optical electrophysiology.

While there are a few examples of this functionality in M-CELS currently, it is not difficult to envision cases where M-CELS could be engineered to sense their environment-either internal or external (see Sec. V B)-then respond via neural bioelectrical output or actuation. One example of this functionality would be stress reporting. It is well known that neural circuits exhibit distinct responses under stress conditions in vivo, ^{310,311} and while the in vivo environment is more complex and involves interactions with many different systems and tissues, organoids, and M-CELS are continuously becoming more complex and beginning to model these interactions,³¹² suggesting the capacity to recapitulate these stress responses, as well as other responses to the environment or other signals. In a widely publicized study,³¹³ observed oscillatory activity in cerebral organoids that closely mimicked patterns characteristic of the human neonatal brain. These, and similar, studies support the assertion that neural organoids are more accurately representing in vivo neural activity and may soon accurately exhibit complex physiological responses (e.g., stress response). As these responses are better understood, they can be monitored and used to observe information on M-CELS' states or engineered for specific downstream actuation.

In addition to neural cells, it is now accepted that non-excitable cells-not just neurons and muscles-sense and generate electrical signals, using bioelectrical signaling to control and coordinate various functions, especially developmental patterning.³¹⁴ While much of the work in this area has been focused on developmental processes and morphogenesis, bioelectrical signaling represents an alternative avenue for engineering and design of M-CELS. With a better understanding, cellular bioelectric properties could be manipulated to control M-CELS' function, providing an additional mode of actuation and output. For example, a pathway involving genetic regulation could be engineered with an upstream bioelectric reporter that could be used to detect pathway activation significantly faster-potentially on the order of seconds-than waiting for genetic changes on the order of hours. This approach would allow for better external readouts on pathway activation and internal states, as well as provide an avenue for the M-CELS itself to sense these states and respond in a specific engineered manner in addition to the inherent response (in this case, gene activation).

5. Structural

Tissues consist of cells within and lining the inner surface of an extracellular matrix that provides a porous structural support and molecular cues that enable cells to adapt to changing conditions. Two key features during development are self-assembly and changes to the microenvironment. Examples include self-organization of endothelial cells and smooth muscle cells to form a vessel *in vitro*¹³⁴ and using the synNotch circuit to promote differential adhesion between cells and produce mulitlayer tissue-like structures.⁵⁵

Expanding the repertoire of molecules in the extracellular matrix can enable programming of M-CELS to synthesize a matrix that meets specific functional needs. The mechanical strength of DNA hydrogels can be adjusted by variation of DNA sequence or creating branching structures.¹⁶⁹ Aptamers could be synthesized attached to long DNA strands to enable specific binding of proteins or peptides. Single stranded DNA can be synthesized to enable the production of bulk amounts for nanocomposites.³¹⁵ By using metabolic processes to regulate DNA synthesis in the extracellular media, different DNA structures can be produced. By coding the variation of metabolic processes, the DNA structures can undergo locomotion.³¹⁵ Such an approach broadens the way in which extracellular synthesis can be used to create M-CELS for novel applications.

VI. OUTLOOK

We have outlined the design rules for building multicellular tissues that leverage coordinated communication between component and cells of different types, functions, and stages of maturity. These design principles outline strategies for M-CELS' design, synthesis, maintenance, and modulation and have been discussed in the context of organoid disease models, implantable chemical factories, and biologically powered robots. These emerging applications support the underlying premise of M-CELS by showing themselves capable of selfassembly, homeostatic maintenance, and dynamic programmable adaptation to changing environmental signals. Moreover, they showcase key advances of how sensing and information processing in M-CELS can drive diverse behaviors, such as mechanical, chemical, optical, bioelectrical, and structural output functions. M-CELS with defined output responses, such as actuation, can be engineered via self-organization of groups of stem cells into mechanically functional tissue (Sec. II), top-down defined co-cultures of different cell types (Sec. III), or some combination of these approaches. Illustrative examples of M-CELS that incorporate the various design principles described in this Review are outlined below, representing the first step toward engineering such systems with complex multifunctional behaviors.

Recent advances in neuromuscular organoid synthesis exemplify the design principles outlined in this Review to generate functional, physiological structures via the assembly of separate modalities. Specifically, Andersen *et al.*³¹⁶ generated human cortico-motor assembloids that accurately model the human corticospinal motor tract by combining three parts: human cortical spheroids, human hindbrain/ cervical spinal cord spheroids, and human skeletal muscle. To achieve this, the authors first utilize design principles regarding development and cell differentiation (Sec. II) to separately develop these regionspecific spheroids. Following this, principles of multi-cellular module integration (Sec. III) are employed to successfully generate corticospinal assembloids, neuromuscular assembloids, and ultimately corticospinal motor assembloids, functionally integrating all three modules. Notably, these assembloids measuring approximately hundreds of micrometers in diameter were able to be maintained and monitored

(Sec. IV) for up to 10 weeks in culture post-assembly and demonstrated functional mechanical output (Sec. V) via muscle contraction in response to optogenetic stimulation of cortical spheroids. As the first example of a completely stem-cell derived neuromuscular tissue capable of functional contraction, these assembloids could offer unprecedented insight into human neuromuscular development and disease. Such sub-millimeter scale systems could have potential nearterm applications in high-throughput biomarker discovery, drug development, and personalized medicine. Future applications of neuromuscular assembloids as completely cellular and externally controllable implants, such as customizable drug delivery pumps, could open up the use of such M-CELS to new realms. Ultimately, these assembloids-which represent a significant advance in complexity and integration over previous 2D or two-component motor unit modelshighlight the promise of the outlined set of design principles for engineering M-CELS with improved applicability.

Certain applications of M-CELS may require larger systems (millimeter-scale and beyond) that also integrate abiotic materials as functional components. Examples of M-CELS in biohybrid robotics highlight how a different fabrication approach can be used to satisfy these parameters. Using many of the design principles outlined in this Review, Raman et al. created muscle actuators with the capacity to heal after injury, and a capability that leverages the adaptive nature of M-CELS and has never been observed in abiotic actuators.¹⁸⁸ First, the authors created biohybrid robots with tissue engineered skeletal muscle (Sec. II) coupled to 3D-printed polymeric scaffolds that helped drive and maintain desired tissue morphology (Sec. III). The muscle actuators were endowed with adaptive response and healing capabilities (Sec. IV) by the inclusion of muscle stem cells that mimicked the satellite cell niche in vivo and via targeted stimulation and exercise of specific regions of the tissue. This approach resulted in full recovery of muscle actuator force (Sec. V) within two days of extensive mechanical damage¹⁹² built on this platform by integrating these muscle actuators with a modular tissue composed of functional motor neurons (Sec. III), leveraging design principles that helped maintain the individual functionalities all the component cells (Sec. IV). These assemblies demonstrated functional mechanical output in response to stimulation of the motor neurons, as with the organoid assembloids described above, showcasing different cell types and design pathways that can lead to M-CELS with similar output responses, depending on the physical size of tissue required and the proposed application. Such systems could be envisioned for use in surgical robotics, such as autonomous muscle-powered grippers that help in performing tasks such as precision suturing. In the longer term, embedding biohybrid robots within exoskeletons could enable deploying them outside of controlled environments, such as in untethered terrain-exploring robots for applications in defense or search and rescue.

While the examples outlined above focus on mechanical actuation, the proposed design principles may be applicable to many other output responses (such as regulated chemical synthesis) and even combinations of output responses. In this respect, their range of uses can be as vast as the imagination of the designer or as numerous as the engineering problems in need of solutions. However, in order to be a useful resource for practitioners of this emerging field, the design principles will require targeted community-driven efforts. Fundamentally, this means refining and updating the principles as new scientific discoveries are made and shared. To do this, the M-CELS' community can mirror other multidisciplinary fields by generating focused symposia to share new information and by creating dedicated training tools and curricula. It can also pursue new methods of engagement by creating databases of design principles or specialized M-CELS' manuscript preprint servers. There has yet to be a collective discussion on how the community can add emerging design principles to the compendium presented in this Review, however, and this remains an important unresolved issue in M-CELS' development and application.

Likewise, to effectively apply M-CELS' technology to solve pressing societal problems, refinement of the design principles will entail thorough integration of ethical and social considerations. As discussed in previous work on the ethics of M-CELS,⁵⁰ this is not a matter of adding ethics onto engineering or pursuing ethics in parallel but rather making implicit value choices more explicit and deciding proactively which avenues of research will have a positive impact. Any M-CELS created, for example, are likely to challenge or complicate traditional notions of biological life and of the human. Prudent choice of cell line (human or non-human), decisions regarding ownership of intellectual property, and careful use of descriptive language ("creature" or "bot"), for example, become especially pressing. Envisioned societal benefits, too, must be subjected to scrutiny to ensure that they are practically feasible-are they still useful when placed in an economic, legal, and cultural context?---and that they represent the actual needs of intended users or beneficiaries. To address these concerns within their work, researchers can draw on the frameworks of "value-sensitive design"³¹ or "responsible research and innovation."318 Thinking beyond the design cycle, it is also crucial that these activities be supported by funding agencies, universities, and other key institutions, such as a special M-CELS' scientific society, funding collaborations with the humanities or social sciences, and motivating broader engagements with policymakers and affected publics.⁵⁰

Looking ahead to the more distant future of M-CELS, these dual challenges (technical and ethical) of curating design principles are not merely obstacles to be overcome. They also represent the transformative aspirations of M-CELS as a field, distinct from pre-existing disciplinary specializations. Researchers have already generated significant collective momentum around this ambitious vision: responsibly design-build-test-debug complex living systems using biological parts. They have crafted new curricula and pedagogical methods for training students in M-CELS³¹⁹ and convened regular interdisciplinary workshops on the ethics and technics of responsible M-CELS research. This community sensibility, ideally, will become a defining and stabilizing feature of the field. Where these trajectories will lead and what they will enable the community to create, however, remain open questions that are worthy of debate. As with the rise of genetics, synthetic biology, and other re-imaginations of biological life, concrete and coordinated community action brings into being new and sometimes unprecedented technological and societal possibilities.

ACKNOWLEDGMENTS

This work was supported by the National Science Foundation Science and Technology Center for Emergent Behaviors of Integrated Cellular Systems, Grant No. 0939511. The authors would like to thank Dr. Shun Zhang, Massachusetts Institute for Technology, for kindly providing the schematic and immunostaining images presented in Fig. 4(c) and Dr. Kristina Davis, Center for Research Computing at the University of Notre Dame for assistance in generating Fig. 2(a).

AUTHOR DECLARATIONS

Conflict of Interests

The authors have no conflicts to disclose.

Ethics Approval

Ethics approval was not required for this article.

Author Contributions

O.A., A.P.P., R.R., S.E.S., and R.P.W. contributed equally to this work.

DATA AVAILABILITY

Data sharing is not applicable to this article as no new data were created or analyzed in this study.

REFERENCES

- ¹J. B. Gurdon, Dev. Biol. 4, 256–273 (1962).
- ²P. Leder and M. W. Nirenberg, Proc. Natl. Acad. Sci. U. S. A. 52(6), 1521–1529 (1964).
- ³H. O. Smith and K. W. Welcox, J. Mol. Biol. 51(2), 379-391 (1970).
- ⁴A. M. Maxam and W. Gilbert, Proc. Natl. Acad. Sci. U. S. A. 74(2), 560–564 (1977).
- ⁵C. A. Hutchison III, S. Phillips, M. H. Edgell, S. Gillam, P. Jahnke, and M. Smith, J. Biol. Chem. 253(18), 6551–6560 (1978).
- ⁶E. Bell, B. Ivarsson, and C. Merrill, Proc. Natl. Acad. Sci. U. S. A. 76(3), 1274–1278 (1979).
- ⁷S. N. Cohen and H. W. Boyer, U.S. Patent No. 4,237,224 (2 December 1980).
- ⁸M. Ptashne, A. Jeffrey, A. D. Johnson, R. Maurer, B. J. Meyer, C. O. Pabo, T. M. Roberts, and R. T. Sauer, Cell **19**(1), 1–11 (1980).
- ⁹D. Bray, Nature **376**(6538), 307–312 (1995).
- ¹⁰H. H. McAdams and L. Shapiro, Science **269**(5224), 650 (1995).
- ¹¹T. D. Chou, S. L. Chen, T. W. Lee, S. G. Chen, T. Y. Cheng, C. H. Lee, T. M.
- Chen, and H. J. Wang, Plast. Reconstr. Surg. 108(2), 378–384 (2001). ¹²I. Wilmut, A. E. Schnieke, J. McWhir, A. J. Kind, and K. H. Campbell, Nature 385(6619), 810–813 (1997).
- ¹³A. Fire, S. Xu, M. K. Montgomery, S. A. Kostas, S. E. Driver, and C. C. Mello, Nature **391**(6669), 806–811 (1998).
- ¹⁴T. S. Gardner, C. R. Cantor, and J. J. Collins, Nature 403(6767), 339–342 (2000).

¹⁵M. B. Elowitz and S. Leibler, Nature **403**(6767), 335–338 (2000).

- ¹⁶M. B. Miller and B. L. Bassler, Annu. Rev. Microbiol. 55(1), 165–199 (2001).
- 17 E. S. Lander, L. M. Linton, B. Birren, C. Nusbaum, M. C. Zody, J. Baldwin, K. Devon, K. Dewar, M. Doyle, W. FitzHugh, R. Funke, D. Gage, K. Harris, A. Heaford, J. Howland, L. Kann, J. Lehoczky, R. LeVine, P. McEwan, K. McKernan, J. Meldrim, J. P. Mesirov, C. Miranda, W. Morris, J. Naylor, C. Raymond, M. Rosetti, R. Santos, A. Sheridan, C. Sougnez, N. Stange-Thomann, N. Stojanovic, A. Subramanian, D. Wyman, J. Rogers, J. Sulston, R. Ainscough, S. Beck, D. Bentley, J. Burton, C. Clee, N. Carter, A. Coulson, R. Deadman, P. Deloukas, A. Dunham, I. Dunham, R. Durbin, L. French, D. Grafham, S. Gregory, T. Hubbard, S. Humphray, A. Hunt, M. Jones, C. Lloyd, A. McMurray, L. Matthews, S. Mercer, S. Milne, J. C. Mullikin, A. Mungall, R. Plumb, M. Ross, R. Shownkeen, S. Sims, R. H. Waterston, R. K. Wilson, L. W. Hillier, J. D. McPherson, M. A. Marra, E. R. Mardis, L. A. Fulton, A. T. Chinwalla, K. H. Pepin, W. R. Gish, S. L. Chissoe, M. C. Wendl, K. D. Delehaunty, T. L. Miner, A. Delehaunty, J. B. Kramer, L. L. Cook, R. S. Fulton, D. L. Johnson, P. J. Minx, S. W. Clifton, T. Hawkins, E. Branscomb, P. Predki, P. Richardson, S. Wenning, T. Slezak, N. Doggett, J.-F. Cheng, A. Olsen, S. Lucas, C. Elkin, E. Uberbacher, M. Frazier, R. A. Gibbs, D. M. Muzny, S. E. Scherer, J. B. Bouck, E. J. Sodergren, K. C. Worley, C. M. Rives, J. H. Gorrell,

M. L. Metzker, S. L. Naylor, R. S. Kucherlapati, D. L. Nelson, G. M. Weinstock, Y. Sakaki, A. Fujiyama, M. Hattori, T. Yada, A. Toyoda, T. Itoh, C. Kawagoe, H. Watanabe, Y. Totoki, T. Taylor, J. Weissenbach, R. Heilig, W. Saurin, F. Artiguenave, P. Brottier, T. Bruls, E. Pelletier, C. Robert, P. Wincker, A. Rosenthal, M. Platzer, G. Nyakatura, S. Taudien, A. Rump, D. R. Smith, L. Doucette-Stamm, M. Rubenfield, K. Weinstock, H. M. Lee, J. Dubois, H. Yang, J. Yu, J. Wang, G. Huang, J. Gu, L. Hood, L. Rowen, A. Madan, S. Qin, R. W. Davis, N. A. Federspiel, A. P. Abola, M. J. Proctor, B. A. Roe, F. Chen, H. Pan, J. Ramser, H. Lehrach, R. Reinhardt, W. R. McCombie, M. de la Bastide, N. Dedhia, H. Blöcker, K. Hornischer, G. Nordsiek, R. Agarwala, L. Aravind, J. A. Bailey, A. Bateman, S. Batzoglou, E. Birney, P. Bork, D. G. Brown, C. B. Burge, L. Cerutti, H.-C. Chen, D. Church, M. Clamp, R. R. Copley, T. Doerks, S. R. Eddy, E. E. Eichler, T. S. Furey, J. Galagan, J. G. R. Gilbert, C. Harmon, Y. Hayashizaki, D. Haussler, H. Hermjakob, K. Hokamp, W. Jang, L. S. Johnson, T. A. Jones, S. Kasif, A. Kaspryzk, S. Kennedy, W. J. Kent, P. Kitts, E. V. Koonin, I. Korf, D. Kulp, D. Lancet, T. M. Lowe, A. McLysaght, T. Mikkelsen, J. V. Moran, N. Mulder, V. J. Pollara, C. P. Ponting, G. Schuler, J. Schultz, G. Slater, A. F. A. Smit, E. Stupka, J. Szustakowki, D. Thierry-Mieg, J. Thierry-Mieg, L. Wagner, J. Wallis, R. Wheeler, A. Williams, Y. I. Wolf, K. H. Wolfe, S.-P. Yang, R.-F. Yeh, F. Collins, M. S. Guyer, J. Peterson, A. Felsenfeld, K. A. Wetterstrand, R. M. Myers, J. Schmutz, M. Dickson, J. Grimwood, D. R. Cox, M. V. Olson, R. Kaul, C. Raymond, N. Shimizu, K. Kawasaki, S. Minoshima, G. A. Evans, M. Athanasiou, R. Schultz, A. Patrinos, M. J. Morgan, and International Human Genome Sequencing Consortium, Nature 409(6822), 860-921 (2001).

- ¹⁸J. Hasty, D. McMillen, and J. J. Collins, Nature **420**(6912), 224–230 (2002).
- ¹⁹R. Weiss and S. Basu, NSC-1: The First Workshop of Non-Silicon Computing (2002).
- ²⁰C. C. Guet, M. B. Elowitz, W. Hsing, and S. Leibler, Science 296(5572), 1466 (2002).
- ²¹J. Cello, A. V. Paul, and E. Wimmer, <u>Science</u> **297**(5583), 1016 (2002).
 - ²²M. R. Atkinson, M. A. Savageau, J. T. Myers, and A. J. Ninfa, Cell 113(5), 597–607 (2003).
 - ²³T. Knight, MIT Synthetic Biology Working Group (2005).
 - ²⁴V. J. J. Martin, D. J. Pitera, S. T. Withers, J. D. Newman, and J. D. Keasling, Nat. Biotechnol. 21(7), 796–802 (2003).
 - ²⁵L. You, R. S. Cox, R. Weiss, and F. H. Arnold, Nature 428(6985), 868–871 (2004).
 - ²⁶S. Basu, Y. Gerchman, C. H. Collins, F. H. Arnold, and R. Weiss, Nature 434(7037), 1130–1134 (2005).
 - ²⁷ A. Levskaya, A. A. Chevalier, J. J. Tabor, Z. B. Simpson, L. A. Lavery, M. Levy, E. A. Davidson, A. Scouras, A. D. Ellington, E. M. Marcotte, and C. A. Voigt, Nature 438(7067), 441–442 (2005).
 - ²⁸K. Takahashi and S. Yamanaka, Cell **126**(4), 663–676 (2006).
 - ²⁹K. Rinaudo, L. Bleris, R. Maddamsetti, S. Subramanian, R. Weiss, and Y. Benenson, Nat. Biotechnol. 25(7), 795–801 (2007).
 - **30**T. L. Deans, C. R. Cantor, and J. J. Collins, Cell **130**(2), 363–372 (2007).
 - ³¹J. Stricker, S. Cookson, M. R. Bennett, W. H. Mather, L. S. Tsimring, and J. Hasty, Nature 456(7221), 516–519 (2008).
 - ³²M. N. Win and C. D. Smolke, Science 322(5900), 456 (2008).
 - ³³H. H. Wang, F. J. Isaacs, P. A. Carr, Z. Z. Sun, G. Xu, C. R. Forest, and G. M. Church, Nature 460(7257), 894–898 (2009).
 - ³⁴J. J. Tabor, H. M. Salis, Z. B. Simpson, A. A. Chevalier, A. Levskaya, E. M. Marcotte, C. A. Voigt, and A. D. Ellington, Cell **137**(7), 1272–1281 (2009).
 - ³⁵D. G. Gibson, J. I. Glass, C. Lartigue, V. N. Noskov, R. Y. Chuang, M. A. Algire, G. A. Benders, M. G. Montague, L. Ma, M. M. Moodie, C. Merryman, S. Vashee, R. Krishnakumar, N. Assad-Garcia, C. Andrews-Pfannkoch, E. A. Denisova, L. Young, Z. Q. Qi, T. H. Segall-Shapiro, C. H. Calvey, P. P. Parmar, C. A. Hutchison III, H. O. Smith, and J. C. Venter, Science **329**(5987), 52–56 (2010).
 - ³⁶X. Song, R. Ju, Y. Xiao, X. Wang, S. Liu, M. Fu, J. Liu, L. Gu, X. Li, and L. Cheng, Int. J. Nanomed. **12**, 7433–7451 (2017).
 - **37**T. Kido and Y. Koui, Methods Mol. Biol. **1905**, 131–142 (2019).
 - ³⁸A. Tamsir, J. J. Tabor, and C. A. Voigt, Nature **469**(7329), 212–215 (2011).
 - ³⁹C. Liu, X. Fu, L. Liu, X. Ren, C. K. L. Chau, S. Li, L. Xiang, H. Zeng, G. Chen, L.-H. Tang, P. Lenz, X. Cui, W. Huang, T. Hwa, and J.-D. Huang, *Science* 334(6053), 238 (2011).

- ⁴⁰J. M. Callura, C. R. Cantor, and J. J. Collins, Proc. Natl. Acad. Sci. U. S. A. 109(15), 5850 (2012).
- ⁴¹A. Prindle, P. Samayoa, I. Razinkov, T. Danino, L. S. Tsimring, and J. Hasty, Nature **481**(7379), 39–44 (2011).
- ⁴²B. Wiedenheft, S. H. Sternberg, and J. A. Doudna, Nature 482(7385), 331–338 (2012).
- ⁴³M. A. Lancaster, M. Renner, C.-A. Martin, D. Wenzel, L. S. Bicknell, M. E. Hurles, T. Homfray, J. M. Penninger, A. P. Jackson, and J. A. Knoblich, Nature 501(7467), 373–379 (2013).

- ⁴⁵N. Annaluru, H. Muller, L. A. Mitchell, S. Ramalingam, G. Stracquadanio, S. M. Richardson, J. S. Dymond, Z. Kuang, L. Z. Scheifele, E. M. Cooper, Y. Cai, K. Zeller, N. Agmon, J. S. Han, M. Hadjithomas, J. Tullman, K. Caravelli, K. Cirelli, Z. Guo, V. London, A. Yeluru, S. Murugan, K. Kandavelou, N. Agier, G. Fischer, K. Yang, J. A. Martin, M. Bilgel, P. Bohutskyi, K. M. Boulier, B. J. Capaldo, J. Chang, K. Charoen, W. J. Choi, P. Deng, J. E. DiCarlo, J. Doong, J. Dunn, J. I. Feinberg, C. Fernandez, C. E. Floria, D. Gladowski, P. Hadidi, I. Ishizuka, J. Jabbari, C. Y. L. Lau, P. A. Lee, S. Li, D. Lin, M. E. Linder, J. Ling, J. Liu, M. London, H. Ma, J. Mao, J. E. McDade, A. McMillan, A. M. Moore, W. C. Oh, Y. Ouyang, R. Patel, M. Paul, L. C. Paulsen, J. Qiu, A. Rhee, M. G. Rubashkin, I. Y. Soh, N. E. Sotuyo, V. Srinivas, A. Suarez, A. Wong, R. Wong, W. R. Xie, Y. Xu, A. T. Yu, R. Koszul, J. S. Bader, J. D. Boeke, and S. Chandrasegaran, <u>Science</u> 344(6179), 55 (2014).
- ⁴⁶J. A. Steinbeck, M. K. Jaiswal, E. L. Calder, S. Kishinevsky, A. Weishaupt, K. V. Toyka, P. A. Goldstein, and L. Studer, Cell Stem Cell 18(1), 134–143 (2016).
- ⁴⁷M. Elowitz and W. A. Lim, Nature 468(7326), 889–890 (2010).
- ⁴⁸V. A. Webster-Wood, O. Akkus, U. A. Gurkan, H. J. Chiel, and R. D. Quinn, Sci. Rob. 2(12), eaap9281 (2017).
- ⁴⁹M. A. Lancaster and M. Huch, Dis. Models Mech. **12**(7), dmm039347 (2019).
- ⁵⁰M. Sample, M. Boulicault, C. Allen, R. Bashir, I. Hyun, M. Levis, C. Lowenthal, D. Mertz, N. Montserrat, M. J. Palmer, K. Saha, and J. Zartman, Biofabrication 11(4), 043001 (2019).
- ⁵¹P. Li, J. S. Markson, S. Wang, S. Chen, V. Vachharajani, and M. B. Elowitz, Science **360**(6388), 543 (2018).
- ⁵²D. Karig, K. M. Martini, T. Lu, N. A. DeLateur, N. Goldenfeld, and R. Weiss, Proc. Natl. Acad. Sci. U. S. A. 115(26), 6572 (2018).
- 53 R. Sekine, T. Shibata, and M. Ebisuya, Nat. Commun. 9(1), 5456 (2018).
- ⁵⁴L. Morsut, K. T. Roybal, X. Xiong, R. M. Gordley, S. M. Coyle, M. Thomson, and W. A. Lim, Cell **164**(4), 780–791 (2016).
- 55S. Toda, L. R. Blauch, S. K. Y. Tang, L. Morsut, and W. A. Lim, Science 361(6398), 156–162 (2018).
- ⁵⁶J. Tordoff and R. Weiss, Nature **559**(7713), 184–185 (2018).
- ⁵⁷M. S. Steinberg and M. Takeichi, Proc. Natl. Acad. Sci. U. S. A. 91(1), 206–209 (1994).
- ⁵⁸E. Cachat, W. Liu, K. C. Martin, X. Yuan, H. Yin, P. Hohenstein, and J. A. Davies, Sci. Rep. 6(1), 20664 (2016).
- ⁵⁹J. Tordoff, M. Krajnc, N. Walczak, M. Lima, J. Beal, S. Shvartsman, and R. Weiss, Cell Rep. Phys. Sci. 2(1), 100305 (2021).
- ⁶⁰A. Warmflash, B. Sorre, F. Etoc, E. D. Siggia, and A. H. Brivanlou, Nat. Methods 11(8), 847–854 (2014).
- ⁶¹D. W. Austin, M. S. Allen, J. M. McCollum, R. D. Dar, J. R. Wilgus, G. S. Sayler, N. F. Samatova, C. D. Cox, and M. L. Simpson, Nature **439**(7076), 608–611 (2006).
- ⁶²G. Chalancon, C. N. J. Ravarani, S. Balaji, A. Martinez-Arias, L. Aravind, R. Jothi, and M. M. Babu, <u>Trends Genet.</u> 28(5), 221–232 (2012).
- 63 A. Becskei and L. Serrano, Nature 405(6786), 590–593 (2000).
- ⁶⁴H. B. Fraser, A. E. Hirsh, G. Giaever, J. Kumm, and M. B. Eisen, PLoS Biol. 2(6), e137 (2004).
- 65 B. Lehner, Mol. Syst. Biol. 4(1), 170 (2008).
- 66 A. Raj and A. van Oudenaarden, Cell 135(2), 216–226 (2008).
- ⁶⁷A. Eldar and M. B. Elowitz, Nature 467(7312), 167-173 (2010).
- ⁶⁸G. Balázsi, A. van Oudenaarden, and J. Collins, Cell 144(6), 910–925 (2011).
- ⁶⁹H. M. Meyer and A. H. K. Roeder, Front. Plant Sci. (published online 2014).
- ⁷⁰C. J. Chan, C.-P. Heisenberg, and T. Hiiragi, Curr. Biol. 27(18), R1024–R1035 (2017).

- ⁷¹J. J. Northey, L. Przybyla, and V. M. Weaver, Cancer Discovery 7(11), 1224–1237 (2017).
- ⁷²C. D. Cox, J. M. McCollum, M. S. Allen, R. D. Dar, and M. L. Simpson, Proc. Natl. Acad. Sci. U. S. A. 105(31), 10809 (2008).
- ⁷³T. Kalmar, C. Lim, P. Hayward, S. Muñoz-Descalzo, J. Nichols, J. Garcia-Ojalvo, and A. M. Arias, PLoS Biol. 7(7), e1000149 (2009).
- ⁷⁴A. Filipczyk, K. Gkatzis, J. Fu, P. Hoppe, H. Lickert, K. Anastassiadis, and T. Schroeder, Cell Stem Cell 13(1), 12–13 (2013).
- ⁷⁵Z. S. Singer, J. Yong, J. Tischler, J. A. Hackett, A. Altinok, M. A. Surani, L. Cai, and M. B. Elowitz, Mol. Cell 55(2), 319–331 (2014).
- **⁷⁶**J. R. Chubb, WIREs Dev. Biol. **6**(6), e284 (2017).
- ⁷⁷R. J. Johnston, Jr. and C. Desplan, Annu. Rev. Cell Dev. Biol. 26, 689–719 (2010).
- **78** R. D. Dar and R. Weiss, APL Bioeng. **2**(2), 020901 (2018).
- ⁷⁹M. M. K. Hansen, W. Y. Wen, E. Ingerman, B. S. Razooky, C. E. Thompson, R. D. Dar, C. W. Chin, M. L. Simpson, and L. S. Weinberger, Cell 173(7), 1609–1621.E15 (2018).
- 80 D. Thieffry, A. M. Huerta, E. Pérez-Rueda, and J. Collado-Vides, Bioessays 20(5), 433-440 (1998).
- ⁸¹M. L. Simpson, C. D. Cox, and G. S. Sayler, Proc. Natl. Acad. Sci. U. S. A. 100(8), 4551 (2003).
- ⁸²C. D. Cox, J. M. McCollum, D. W. Austin, M. S. Allen, R. D. Dar, and M. L. Simpson, Chaos 16(2), 026102 (2006).
- ⁸³J. M. Raser and E. K. Shea, Science **309**(5743), 2010 (2005).
- ⁸⁴M. Kaern, T. C. Elston, W. J. Blake, and J. J. Collins, Nat. Rev. Genet. 6(6), 451–464 (2005).
- ⁸⁵M. L. Simpson, C. D. Cox, M. S. Allen, J. M. McCollum, R. D. Dar, D. K. Karig, and J. F. Cooke, WIREs Nanomed. Nanobiotechnol. 1(2), 214–225 (2009).
- ⁸⁶M. Uyttewaal, A. Burian, K. Alim, B. Landrein, D. Borowska-Wykret, A. Dedieu, A. Peaucelle, M. Ludynia, J. Traas, A. Boudaoud, D. Kwiatkowska, and O. Hamant, Cell **149**(2), 439–451 (2012).
- 87 E. Marinari, A. Mehonic, S. Curran, J. Gale, T. Duke, and B. Baum, Nature 484(7395), 542–545 (2012).
- ⁸⁸A. C. Martin, M. Kaschube, and E. F. Wieschaus, Nature **457**(7228), 495–499 (2009).
- ⁸⁹M. Miller, M. Hafner, E. Sontag, N. Davidsohn, S. Subramanian, P. E. M. Purnick, D. Lauffenburger, and R. Weiss, PLoS Comput. Biol. 8(7), e1002579 (2012).
- ⁹⁰M. Wu, R.-Q. Su, X. Li, T. Ellis, Y.-C. Lai, and X. Wang, Proc. Natl. Acad. Sci. U. S. A. 110(26), 10610 (2013).
- ⁹¹P. Guye, M. R. Ebrahimkhani, N. Kipniss, J. J. Velazquez, E. Schoenfeld, S. Kiani, L. G. Griffith, and R. Weiss, Nat. Commun. 7(1), 10243 (2016).
- ⁹²K. F. Murphy, G. Balázsi, and J. J. Collins, Proc. Natl. Acad. Sci. U. S. A. 104(31), 12726 (2007).
- ⁹³K. F. Murphy, R. M. Adams, X. Wang, G. Balázsi, and J. J. Collins, Nucl. Acids Res. 38(8), 2712–2726 (2010).
- 94 M. T. Guinn and G. Balázsi, Nucl. Acids Res. 47(14), 7703-7714 (2019).
- ⁹⁵R. D. Dar, N. N. Hosmane, M. R. Arkin, R. F. Siliciano, and L. S. Weinberger, Science 344(6190), 1392 (2014).
- ⁹⁶M. R. Megaridis, Y. Lu, E. N. Tevonian, K. M. Junger, J. M. Moy, K. Bohn-Wippert, and R. D. Dar, <u>APL Bioeng</u>, 2(2), 026106 (2018).
- 97K. Bohn-Wippert, E. N. Tevonian, Y. Lu, M.-Y. Huang, M. R. Megaridis, and R. D. Dar, Cell Rep. 25(13), 3844–3857.e3845 (2018).
- ⁹⁸D. P. Doupé, A. M. Klein, B. D. Simons, and P. H. Jones, Dev. Cell 18(2), 317–323 (2010).
- ⁹⁹S. Gómez-López, R. G. Lerner, and C. Petritsch, Cell. Mol. Life Sci. 71(4), 575–597 (2014).
- 100 J.-E. Dietrich and T. Hiiragi, Development 134(23), 4219 (2007).
- ¹⁰¹J. S. Biggins, C. Royer, T. Watanabe, and S. Srinivas, Semin. Cell Dev. Biol. 47-48, 74-79 (2015).
- 102B. Plusa, A. Piliszek, S. Frankenberg, J. Artus, and A.-K. Hadjantonakis, Development 135(18), 3081 (2008).
- 103Y. Yamanaka, F. Lanner, and J. Rossant, Development 137(5), 715 (2010).
- 104 T. Cheng, H. Shen, D. Giokas, J. Gere, D. G. Tenen, and D. T. Scadden, Proc. Natl. Acad. Sci. U. S. A. 93(23), 13158–13163 (1996).

⁴⁴P. Siuti, J. Yazbek, and T. K. Lu, Nat. Biotechnol. 31(5), 448-452 (2013).

- ¹⁰⁵S. A. Jackson, J. Schiesser, E. G. Stanley, and A. G. Elefanty, PLoS One 5(5), e10706 (2010).
- 106 R. Chaerkady, C. L. Kerr, A. Marimuthu, D. S. Kelkar, M. K. Kashyap, M. Gucek, J. D. Gearhart, and A. Pandey, J. Proteome Res. 8(3), 1315–1326 (2009).
- 107 S. G. Almalki and D. K. Agrawal, Differentiation **92**(1–2), 41–51 (2016).
- ¹⁰⁸K. Yamamizu, Y. Piao, A. Sharov, V. Zsiros, H. Yu, K. Nakazawa, D. Schlessinger, and M. S. H. Ko, Stem Cell Rep. 1(6), 545–559 (2013).
- 109 A. Pfefferle, H. Netskar, E. H. Ask, S. Lorenz, J. P. Goodridge, E. Sohlberg, T. Clancy, and K.-J. Malmberg, "A temporal transcriptional map of human natural killer cell differentiation," bioRxiv:630657 (2019).
- ¹¹⁰E. Becht, L. McInnes, J. Healy, C.-A. Dutertre, I. W. H. Kwok, L. G. Ng, F. Ginhoux, and E. W. Newell, Nat. Biotechnol. 37(1), 38–44 (2019).
- ^{III}L. Haghverdi, F. Buettner, and F. J. Theis, <u>Bioinformatics</u> 31(18), 2989–2998 (2015).
- ¹¹²K. Takahashi, K. Tanabe, M. Ohnuki, M. Narita, T. Ichisaka, K. Tomoda, and S. Yamanaka, Cell 131(5), 861–872 (2007).
- ¹¹³H. Nakajima, Keio J. Med. **60**(2), 47–55 (2011).
- ¹¹⁴R. L. Davis, H. Weintraub, and A. B. Lassar, Cell **51**(6), 987–1000 (1987).
- ¹¹⁵T. Ra'em and S. Cohen, Adv. Biochem. Eng. Biotechnol. **126**, 227–262 (2011).
- ¹¹⁶A. W. Lund, B. Yener, J. P. Stegemann, and G. E. Plopper, Tissue Eng., Part B 15(3), 371–380 (2009).
- ¹¹⁷Y. Li, X. Zhong, J. Yan, J. Lin, S. Tang, X. Wu, S. Li, G. Feng, and Y. Yi, Yan Ke Xue Bao 18(1), 37–44 (2002).
- ¹¹⁸Y. Li, J. Ge, J. Yan, X. Zhong, Y. Guo, S. Li, X. Wu, J. Ling, G. Feng, J. Zhang, and Y. Yi, Yan Ke Xue Bao 15(3), 131–136 (1999).
- ¹¹⁹A. Bertero, S. Brown, P. Madrigal, A. Osnato, D. Ortmann, L. Yiangou, J. Kadiwala, N. C. Hubner, I. R. de Los Mozos, C. Sadée, A.-S. Lenaerts, S. Nakanoh, R. Grandy, E. Farnell, J. Ule, H. G. Stunnenberg, S. Mendjan, and L. Vallier, Nature 555(7695), 256–259 (2018).
- ¹²⁰M. Rodrigues, L. G. Griffith, and A. Wells, <u>Stem Cell Res. Ther.</u> 1(4), 32–32 (2010).
- ¹²¹A. C. Mullen and J. L. Wrana, Cold Spring Harbor Perspect. Biol. 9(7), a022186 (2017).
- ¹²²M. Amit, M. K. Carpenter, M. S. Inokuma, C.-P. Chiu, C. P. Harris, M. A. Waknitz, J. Itskovitz-Eldor, and J. A. Thomson, Dev. Biol. **227**(2), 271–278 (2000).
- ¹²³L. Wang, T. C. Schulz, E. S. Sherrer, D. S. Dauphin, S. Shin, A. M. Nelson, C. B. Ware, M. Zhan, C.-Z. Song, X. Chen, S. N. Brimble, A. McLean, M. J. Galeano, E. W. Uhl, K. A. D'Amour, J. D. Chesnut, M. S. Rao, C. A. Blau, and A. J. Robins, Blood 110(12), 4111–4119 (2007).
- ¹²⁴E. Hannezo and C. P. Heisenberg, Cell 178(1), 12–25 (2019).
- ¹²⁵J. A. Brassard and M. P. Lutolf, Cell Stem Cell 24(6), 860–876 (2019).
- ¹²⁶E. Garreta, R. D. Kamm, S. M. Chuva de Sousa Lopes, M. A. Lancaster, R. Weiss, X. Trepat, I. Hyun, and N. Montserrat, Nat. Mater. 20(2), 145–155 (2021).
- ¹²⁷N. Arora, J. Imran Alsous, J. W. Guggenheim, M. Mak, J. Munera, J. M. Wells, R. D. Kamm, H. H. Asada, S. Y. Shvartsman, and L. G. Griffith, Development 144(6), 1128–1136 (2017).
- ¹²⁸S. M. Czerniecki, N. M. Cruz, J. L. Harder, R. Menon, J. Annis, E. A. Otto, R. E. Gulieva, L. V. Islas, Y. K. Kim, L. M. Tran, T. J. Martins, J. W. Pippin, H. Fu, M. Kretzler, S. J. Shankland, J. Himmelfarb, R. T. Moon, N. Paragas, and B. S. Freedman, Cell Stem Cell **22**(6), 929–940.E4 (2018).
- ¹²⁹A. Manfrin, Y. Tabata, E. R. Paquet, A. R. Vuaridel, F. R. Rivest, F. Naef, and M. P. Lutolf, Nat. Methods 16(7), 640–648 (2019).
- ¹³⁰E. Berger, C. Magliaro, N. Paczia, A. S. Monzel, P. Antony, C. L. Linster, S. Bolognin, A. Ahluwalia, and J. C. Schwamborn, Lab Chip 18(20), 3172–3183 (2018).
- ¹³¹T. DiStefano, H. Y. Chen, C. Panebianco, K. D. Kaya, M. J. Brooks, L. Gieser, N. Y. Morgan, T. Pohida, and A. Swaroop, Stem Cell Rep. **10**(1), 300–313 (2018).
- 132 N. Gjorevski, N. Sachs, A. Manfrin, S. Giger, M. E. Bragina, P. Ordóñez-Morán, H. Clevers, and M. P. Lutolf, Nature 539(7630), 560–564 (2016).
- 133 R. Raman and R. Langer, Adv. Mater. 32(13), 1901969 (2020).
- ¹³⁴L. Andrique, G. Recher, K. Alessandri, N. Pujol, M. Feyeux, P. Bon, L. Cognet, P. Nassoy, and A. Bikfalvi, Sci. Adv. 5(6), eaau6562 (2019).

- ¹³⁵H. J. Kim, D. Huh, G. Hamilton, and D. E. Ingber, Lab Chip 12(12), 2165–2174 (2012).
- ¹³⁶M. Kasendra, A. Tovaglieri, A. Sontheimer-Phelps, S. Jalili-Firoozinezhad, A. Bein, A. Chalkiadaki, W. Scholl, C. Zhang, H. Rickner, C. A. Richmond, H. Li, D. T. Breault, and D. E. Ingber, Sci. Rep. 8(1), 2871 (2018).
- ¹³⁷Y. I. Wang, H. E. Abaci, and M. L. Shuler, Biotechnol. Bioeng. 114(1), 184–194 (2017).
- ¹³⁸M. L. Suster and M. Bate, Nature **416**(6877), 174–178 (2002).
- ¹³⁹M. Campisi, Y. Shin, T. Osaki, C. Hajal, V. Chiono, and R. D. Kamm, Biomaterials 180, 117–129 (2018).
- ¹⁴⁰E. H. Neal, N. A. Marinelli, Y. Shi, P. M. McClatchey, K. M. Balotin, D. R. Gullett, K. A. Hagerla, A. B. Bowman, K. C. Ess, J. P. Wikswo, and E. S. Lippmann, Stem Cell Rep. 12(6), 1380–1388 (2019).
- ¹⁴¹A. H. Huang, Dev. Biol. **429**(2), 420–428 (2017).
- ¹⁴²T. K. Merceron, M. Burt, Y. J. Seol, H. W. Kang, S. J. Lee, J. J. Yoo, and A. Atala, Biofabrication 7(3), 035003 (2015).
- ¹⁴³L. M. Larkin, S. Calve, T. Y. Kostrominova, and E. M. Arruda, Tissue Eng. 12(11), 3149–3158 (2006).
- 144 T. Osaki, V. Sivathanu, and R. D. Kamm, Biomaterials 156, 65–76 (2018).
- ¹⁴⁵Y. Oh, G. S. Cho, Z. Li, I. Hong, R. Zhu, M. J. Kim, Y. J. Kim, E. Tampakakis, L. Tung, R. Huganir, X. Dong, C. Kwon, and G. Lee, Cell Stem Cell 19(1), 95–106 (2016).
- ¹⁴⁶N. R. Martin, S. L. Passey, D. J. Player, V. Mudera, K. Baar, L. Greensmith, and M. P. Lewis, Tissue Eng., Part A 21(19–20), 2595–2604 (2015).
- ¹⁴⁷O. Aydin, A. P. Passaro, M. Elhebeary, G. J. Pagan-Diaz, A. Fan, S. Nuethong, R. Bashir, S. L. Stice, and M. T. A. Saif, APL Bioeng. 4(1), 016107 (2020).
- ¹⁴⁸G. Cao and C. P. Ko, J. Neurosci. 27(25), 6712–6722 (2007).
- ¹⁴⁹A. Tourovskaia, T. F. Kosar, and A. Folch, Biophys. J. **90**(6), 2192–2198 (2006).
- 150 B. G. Zhang, A. F. Quigley, J. L. Bourke, C. J. Nowell, D. E. Myers, P. F. Choong, and R. M. Kapsa, Dev. Neurobiol. 76(5), 551–565 (2016).
- ¹⁵¹H. B. Peng, J. F. Yang, Z. Dai, C. W. Lee, H. W. Hung, Z. H. Feng, and C. P. Ko, J. Neurosci. 23(12), 5050–5060 (2003).
- 152V. Nurcombe, M. A. Hill, K. L. Eagleson, and M. R. Bennett, Brain Res. 291(1), 19–28 (1984).
- ¹⁵³M. A. Fox, J. R. Sanes, D. B. Borza, V. P. Eswarakumar, R. Fässler, B. G. Hudson, S. W. John, Y. Ninomiya, V. Pedchenko, S. L. Pfaff, M. N. Rheault, Y. Sado, Y. Segal, M. J. Werle, and H. Umemori, Cell **129**(1), 179–193 (2007).
- ¹⁵⁴M. Weitkunat, A. Kaya-Çopur, S. Grill, and F. Schnorrer, Curr. Biol. 24(7), 705–716 (2014).
- 155 S. G. M. Uzel, R. J. Platt, V. Subramanian, T. M. Pearl, C. J. Rowlands, V. Chan, L. A. Boyer, P. T. C. So, and R. D. Kamm, Sci. Adv. 2(8), e1501429 (2016).
- 156 T. Osaki, S. G. M. Uzel, and R. D. Kamm, Sci. Adv. 4(10), eaat5847 (2018).
- ¹⁵⁷X. Wang, M. Cirit, J. S. Wishnok, L. G. Griffith, and S. R. Tannenbaum, Anal. Chem. **91**(13), 8667–8675 (2019).
- ¹⁵⁸R. Raman, C. Cvetkovic, and R. Bashir, Nat. Protoc. **12**(3), 519–533 (2017).

- ¹⁶⁰R. Raman, B. Bhaduri, M. Mir, A. Shkumatov, M. K. Lee, G. Popescu, H. Kong, and R. Bashir, Adv. Healthcare Mater. 5(5), 610–619 (2016).
- ¹⁶¹J. S. Miller, K. R. Stevens, M. T. Yang, B. M. Baker, D.-H. T. Nguyen, D. M. Cohen, E. Toro, A. A. Chen, P. A. Galie, X. Yu, R. Chaturvedi, S. N. Bhatia, and C. S. Chen, Nat. Mater. 11(9), 768–774 (2012).
- ¹⁶²M. A. Skylar-Scott, S. G. M. Uzel, L. L. Nam, J. H. Ahrens, R. L. Truby, S. Damaraju, and J. A. Lewis, Sci. Adv. 5(9), eaaw2459 (2019).
- ¹⁶³G.-Z. Yang, J. Bellingham, P. E. Dupont, P. Fischer, L. Floridi, R. Full, N. Jacobstein, V. Kumar, M. McNutt, R. Merrifield, B. J. Nelson, B. Scassellati, M. Taddeo, R. Taylor, M. Veloso, Z. L. Wang, and R. Wood, Sci. Rob. 3(14), eaar7650 (2018).
- ¹⁶⁴H. G. Yi, H. Lee, and D. W. Cho, Bioengineering 4(1), 10 (2017).
- ¹⁶⁵D. Scaini and L. Ballerini, Curr. Opin. Neurobiol. **50**, 50–55 (2018).
- ¹⁶⁶O. Aydin, X. Zhang, S. Nuethong, G. J. Pagan-Diaz, R. Bashir, M. Gazzola, and M. T. A. Saif, Proc. Natl. Acad. Sci. U. S. A. 116(40), 19841–19847 (2019).
- ¹⁶⁷N. W. Xu and J. O. Dabiri, Sci. Adv. 6(5), eaaz3194 (2020).
- ¹⁶⁸R. Raman and R. Bashir, Adv. Healthcare Mater. 6(20), 1700496 (2017).

¹⁵⁹M. R. Blatchley, F. Hall, S. Wang, H. C. Pruitt, and S. Gerecht, Sci. Adv. 5(3), eaau7518 (2019).

- ¹⁶⁹Q. Li, K. Nan, P. L. Floch, Z. Lin, H. Sheng, and J. Liu, "Cyborg organoids: Implantation of nanoelectronics via organogenesis for tissue-wide electrophysiology," Nano Lett. **19**(8), 5781–5789 (2019).
- ¹⁷⁰L. G. Griffith and M. A. Swartz, Nat. Rev. Mol. Cell Biol. 7(3), 211-224 (2006).
- ¹⁷⁷J. Michl, K. C. Park, and P. Swietach, Commun. Biol. 2(1), 144 (2019).
- ¹⁷²P. J. Hung, P. J. Lee, P. Sabounchi, R. Lin, and L. P. Lee, Biotechnol. Bioeng. 89(1), 1–8 (2005).
- 173 B. P. Weegman, P. Nash, A. L. Carlson, K. J. Voltzke, Z. Geng, M. Jahani, B. B. Becker, K. K. Papas, and M. T. Firpo, PLoS One 8(10), e76611 (2013).
- ¹⁷⁴J.-M. Bielser, M. Wolf, J. Souquet, H. Broly, and M. Morbidelli, Biotechnol. Adv. 36(4), 1328–1340 (2018).
- ¹⁷⁵J. D. Humphrey, E. R. Dufresne, and M. A. Schwartz, Nat. Rev. Mol. Cell Biol. 15(12), 802–812 (2014).
- 176 A. J. Ford and P. Rajagopalan, WIREs Nanomed. Nanobiotechnol. 10(4), e1503 (2018).
- ¹⁷⁷D. E. Discher, D. J. Mooney, and P. W. Zandstra, Science 324(5935), 1673–1677 (2009).
- ¹⁷⁸A. J. Engler, S. Sen, H. L. Sweeney, and D. E. Discher, Cell **126**(4), 677–689 (2006).
- ¹⁷⁹C. Frantz, K. M. Stewart, and V. M. Weaver, J. Cell Sci. **123**(24), 4195 (2010).
- 180 M. C. Ferrall-Fairbanks, D. M. West, S. A. Douglas, R. D. Averett, and M. O. Platt, Protein Sci. 27(3), 714–724 (2018).
- ¹⁸¹S. A. Douglas, S. E. LaMothe, T. S. Singleton, R. Averett, and M. O. Platt, FASEB J. **32**(S1), 143.6–143.6 (2018).
- ¹⁸²S. A. Douglas, S. E. Lamothe, T. S. Singleton, R. D. Averett, and M. O. Platt, Biochim. Biophys. Acta, Gen. Sub. **1862**(9), 1925–1932 (2018).
- ¹⁸³K. J. Regehr, M. Domenech, J. T. Koepsel, K. C. Carver, S. J. Ellison-Zelski, W. L. Murphy, L. A. Schuler, E. T. Alarid, and D. J. Beebe, Lab Chip 9(15), 2132–2139 (2009).
- ¹⁸⁴D. Park, J. Lim, J. Y. Park, and S.-H. Lee, Stem Cells Transl. Med. 4(11), 1352–1368 (2015).
- ¹⁸⁵S. Allazetta and M. P. Lutolf, Curr. Opin. Biotechnol. 35, 86–93 (2015).
- ¹⁸⁶M. Karimi, S. Bahrami, H. Mirshekari, S. M. M. Basri, A. B. Nik, A. R. Aref, M. Akbari, and M. R. Hamblin, Lab Chip **16**(14), 2551–2571 (2016).
- ¹⁸⁷M. Juhas, G. C. Engelmayr, A. N. Fontanella, G. M. Palmer, and N. Bursac, Proc. Natl. Acad. Sci. U. S. A. **111**(15), 5508 (2014).
- ¹⁸⁸R. Raman, L. Grant, Y. Seo, C. Cvetkovic, M. Gapinske, A. Palasz, H. Dabbous, H. Kong, P. P. Pinera, and R. Bashir, Adv. Healthcare Mater. 6(12), 1700030 (2017).
- ¹⁸⁹P.-X. Zhang, N. Han, Y.-H. Kou, Q.-T. Zhu, X.-L. Liu, D.-P. Quan, J.-G. Chen, and B.-G. Jiang, Neural Regener. Res. 14(1), 51–58 (2019).
- 190 L. Ricotti, B. Trimmer, A. W. Feinberg, R. Raman, K. K. Parker, R. Bashir, M. Sitti, S. Martel, P. Dario, and A. Menciassi, Sci. Rob. 2(12), eaaq0495 (2017).
- ¹⁹¹R. Raman, C. Cvetkovic, S. G. M. Uzel, R. J. Platt, P. Sengupta, R. D. Kamm, and R. Bashir, Proc. Natl. Acad. Sci. U. S. A. 113(13), 3497 (2016).
- ¹⁹²C. Cvetkovic, R. Raman, V. Chan, B. J. Williams, M. Tolish, P. Bajaj, M. S. Sakar, H. H. Asada, M. T. A. Saif, and R. Bashir, Proc. Natl. Acad. Sci. U. S. A. 111(28), 10125 (2014).
- ¹⁹³B. Koo, B. Choi, H. Park, and K. J. Yoon, Mol. Cells **42**(9), 617–627 (2019).
- ¹⁹⁴G. Quadrato, T. Nguyen, E. Z. Macosko, J. L. Sherwood, S. Min Yang, D. R. Berger, N. Maria, J. Scholvin, M. Goldman, J. P. Kinney, E. S. Boyden, J. W. Lichtman, Z. M. Williams, S. A. McCarroll, and P. Arlotta, Nature 545(7652), 48–53 (2017).
- ¹⁹⁵D. Poli, C. Magliaro, and A. Ahluwalia, Front. Neurosci. 13, 162 (2019).
- ¹⁹⁶S. P. Paşca, Nature 553(7689), 437–445 (2018).
- ¹⁹⁷H. Wang, Front. Synaptic Neurosci. 10, 15 (2018).
- ¹⁹⁸G. J. Pagan-Diaz, J. Drnevich, K. P. Ramos-Cruz, R. Sam, P. Sengupta, and R. Bashir, Sci. Rep. 10(1), 12460 (2020).
- ¹⁹⁹A. Bhaduri, M. G. Andrews, W. Mancia Leon, D. Jung, D. Shin, D. Allen, D. Jung, G. Schmunk, M. Haeussler, J. Salma, A. A. Pollen, T. J. Nowakowski, and A. R. Kriegstein, Nature 578(7793), 142–148 (2020).
- ²⁰⁰S. Takamiya, S. Yuki, J. Hirokawa, H. Manabe, and Y. Sakurai, Neurosci. Res. 153, 22–26 (2020).
- ²⁰¹T. Stieglitz, H. H. Ruf, M. Gross, M. Schuettler, and J. U. Meyer, Biosens. Bioelectron. 17(8), 685–696 (2002).

- ²⁰²J. Liu, Biomimetics through Nanoelectronics: Development of Three Dimensional Macroporous Nanoelectronics for Building Smart Materials, Cyborg Tissues and Injectable Biomedical Electronics (Springer, 2017).
- ²⁰³G. J. Pagan-Diaz, K. P. Ramos-Cruz, R. Sam, M. E. Kandel, O. Aydin, M. T. A. Saif, G. Popescu, and R. Bashir, Proc. Natl. Acad. Sci. U. S. A. **116**(51), 25932 (2019).
- 204 F. D. Broccard, S. Joshi, J. Wang, and G. Cauwenberghs, J. Neural Eng. 14(4), 041002 (2017).
- ²⁰⁵S. T. Keene, C. Lubrano, S. Kazemzadeh, A. Melianas, Y. Tuchman, G. Polino, P. Scognamiglio, L. Cinà, A. Salleo, Y. van de Burgt, and F. Santoro, Nat. Mater. **19**(9), 969–973 (2020).
- 206 H. Keren, J. Partzsch, S. Marom, and C. G. Mayr, Front. Neurosci. 13, 432 (2019).
- 207 K. C. Rustad, M. Sorkin, B. Levi, M. T. Longaker, and G. C. Gurtner, Organogenesis 6(3), 151–157 (2010).
- 208 B. P. Chan and K. W. Leong, Eur. Spine J. 17(Suppl 4), 467–479 (2008).
- ²⁰⁹T. Reinecke, P. Biechele, V. Schulte, T. Scheper, and S. Zimmermann, Procedia Eng. **120**, 548–551 (2015).
- ²¹⁰M. F. Miniotis, A. Mukwaya, and A. G. Wingren, PLoS One 9(9), e106546 (2014).
- ²¹⁰M. Mir, T. Kim, A. Majumder, M. Xiang, R. Wang, S. C. Liu, M. U. Gillette, S. Stice, and G. Popescu, Sci. Rep. 4(1), 4434 (2014).
- ²¹²K. Kim and W. R. Wagner, Ann. Biomed. Eng. 44(3), 621–635 (2016).
- 213 S. F. Othman, E. T. Curtis, S. A. Plautz, A. K. Pannier, S. D. Butler, and H. Xu, NMR Biomed. 25(3), 452–463 (2012).
- ²¹⁴L. Tian, S. A. Hires, T. Mao, D. Huber, M. E. Chiappe, S. H. Chalasani, L. Petreanu, J. Akerboom, S. A. McKinney, E. R. Schreiter, C. I. Bargmann, V. Jayaraman, K. Svoboda, and L. L. Looger, Nat. Methods 6(12), 875–881 (2009).
- ²¹⁵V. Ruprecht, P. Monzo, A. Ravasio, Z. Yue, E. Makhija, P. O. Strale, N. Gauthier, G. V. Shivashankar, V. Studer, C. Albiges-Rizo, and V. Viasnoff, J. Cell Sci. **130**(1), 51–61 (2017).
- ²¹⁶M. A. Lemmon and J. Schlessinger, Cell 141(7), 1117–1134 (2010).
- 217 A. I. Ségaliny, M. Tellez-Gabriel, M.-F. Heymann, and D. Heymann, J. Bone Oncol. 4(1), 1–12 (2015).
- ²¹⁸X. Wang, P. Lupardus, S. L. Laporte, and K. C. Garcia, <u>Annu. Rev. Immunol.</u> 27, 29–60 (2009).
- ²¹⁹A. J. Brooks, F. Dehkhoda, and B. B. Kragelund, in *Principles of Endocrinology and Hormone Action*, edited by A. Belfiore and D. LeRoith (Springer International Publishing, Cham, 2018), pp. 157–185.
- ²²⁰N. Tuteja, Plant Signaling Behav. 4(10), 942–947 (2009).
- ²²¹J. Z. Kechagia, J. Ivaska, and P. Roca-Cusachs, Nat. Rev. Mol. Cell Biol. 20(8), 457–473 (2019).
- ²²²N. Unwin, Cell **72**, 31–41 (1993).
- 223 A. Samanta, T. E. T. Hughes, and V. Y. Moiseenkova-Bell, Subcell. Biochem. 87, 141–165 (2018).
- ²²⁴ TRP Ion Channel Function in Sensory Transduction and Cellular Signaling Cascades, edited by W. B. Liedtke and S. Heller (CRC Press/Taylor & Francis Group LLC, Boca Raton, FL, 2007).
- 225 C. R. Chillakuri, D. Sheppard, S. M. Lea, and P. A. Handford, Semin. Cell Dev. Biol. 23(4), 421–428 (2012).
- 226 G. Gaud, R. Lesourne, and P. E. Love, Nat. Rev. Immunol. 18(8), 485–497 (2018).
- **227** B. Treanor, Immunology **136**(1), 21–27 (2012).
- ²²⁸G. P. Amarante-Mendes, S. Adjemian, L. M. Branco, L. C. Zanetti, R. Weinlich, and K. R. Bortoluci, Front. Immunol. 9, 2379 (2018).
- ²²⁹R. M. Evans, Science 240(4854), 889 (1988).
- ²³⁰J. W. Pike and M. B. Meyer, Endocrinol. Metab. Clin. North Am. **39**(2), 255–269 (2010).
- ²³¹P. Huang, V. Chandra, and F. Rastinejad, Chem. Rev. 114(1), 233–254 (2014).
 ²³²P. Balaguer, V. Delfosse, M. Grimaldi, and W. Bourguet, C. R. Biol. 340(9),
- 414–420 (2017).
 233 C. Graham, R. Hewitson, A. Pagliuca, and R. Benjamin, Clin. Med. 18(4),
- 324-328 (2018).
- 234 E. S. Boyden, F. Zhang, E. Bamberg, G. Nagel, and K. Deisseroth, Nat. Neurosci. 8(9), 1263–1268 (2005).

- 235 M. Hicks, T. T. Bachmann, and B. Wang, ChemPhysChem 21(2), 132–144 (2020).
- **236**S. Firestein, Nature **413**(6852), 211–218 (2001).
- ²³⁷A. J. Hudspeth, Nat. Rev. Neurosci. 15(9), 600-614 (2014).
- 238 E. G. Ekdale, J. Anat. 228(2), 324-337 (2016).
- 239J. C. Tuthill and E. Azim, Curr. Biol. 28(5), R194-R203 (2018).
- ²⁴⁰T. D. Lamb, Eye **30**(2), 179–185 (2016).
- ²⁴¹R. Singh, O. Cuzzani, F. Binette, H. Sternberg, M. D. West, and I. O. Nasonkin, Stem Cell Rev. Rep. 14(4), 463–483 (2018).
- ²⁴²S. S. Fong, A. R. Joyce, and B. Ø. Palsson, Genome Res. 15(10), 1365–1372 (2005).
- 243 A. R. Joyce and B. Palsson, Methods Mol. Biol. 416, 433–457 (2008).
- ²⁴⁴E. Fischer and U. Sauer, Nat. Genet. **37**(6), 636–640 (2005).
- ²⁴⁵M. L. Goldberg and C. G. Biava, Biochim. Biophys. Acta 454(3), 457–468 (1976).
- ²⁴⁶P. Jolicoeur, A. Lemay, F. Labrie, and A. L. Steiner, Exp. Cell Res. 89(2), 231–240 (1974).
- 247 A. Efeyan, W. C. Comb, and D. M. Sabatini, Nature 517(7534), 302–310 (2015).
- ²⁴⁸Y.-P. Wang and Q.-Y. Lei, Signal Transduction Targeted Ther. 3(1), 30 (2018).
- ²⁴⁹Y. Xiong, M. McCormack, L. Li, Q. Hall, C. Xiang, and J. Sheen, Nature 496(7444), 181–186 (2013).
- ²⁵⁰M. Fried and D. M. Crothers, Nucl. Acids Res. 9(23), 6505–6525 (1981).
- ²⁵¹E. M. Ozbudak, M. Thattai, H. N. Lim, B. I. Shraiman, and A. van Oudenaarden, Nature **427**(6976), 737–740 (2004).
- ²⁵²R. A. Cairns, I. S. Harris, and T. W. Mak, Nat. Rev. Cancer 11(2), 85–95 (2011).
- **253**E. Dekel and U. Alon, Nature **436**(7050), 588–592 (2005).
- 254 P. Chambon, J. D. Weill, and P. Mandel, Biochem. Biophys. Res. Commun. 11(1), 39–43 (1963).
- 255 R. J. Collier and A. M. Pappenheimer, Jr., J. Exp. Med. 120(6), 1019–1039 (1964).
- 256 T. Honjo, Y. Nishizuka, and O. Hayaishi, J. Biol. Chem. 243(12), 3553–3555 (1968).
- **257** A. Aranda and A. Pascual, Physiol. Rev. **81**(3), 1269–1304 (2001).
- ²⁵⁸J. A. Wilhelm and K. S. McCarty, Cancer Res. **30**(2), 418–425 (1970).
- ²⁵⁹S. B. Rothbart and B. D. Strahl, Biochim. Biophys. Acta 1839(8), 627–643 (2014).
- ²⁶⁰X. Gao, S.-H. Lin, F. Ren, J.-T. Li, J.-J. Chen, C.-B. Yao, H.-B. Yang, S.-X. Jiang, G.-Q. Yan, D. Wang, Y. Wang, Y. Liu, Z. Cai, Y.-Y. Xu, J. Chen, W. Yu, P.-Y. Yang, and Q.-Y. Lei, Nat. Commun. 7(1), 11960 (2016).
- 261 R. Bagur and G. Hajnóczky, Mol. Cell 66(6), 780-788 (2017).
- ²⁶² A. C. Newton, M. D. Bootman, and J. D. Scott, Cold Spring Harbor Perspect. Biol. 8(8), a005926 (2016).
- 263]. K. Foskett, C. White, K.-H. Cheung, and D.-O. D. Mak, Physiol. Rev. 87(2), 593–658 (2007).
- ²⁶⁴C. L. Klaips, G. G. Jayaraj, and F. U. Hartl, J. Cell Biol. 217(1), 51–63 (2018).
- **265** J. C. H. Tam and D. A. Jacques, J. Leukocyte Biol. **96**(2), 233–244 (2014).
- ²⁶⁶J. J. Tabor, T. S. Bayer, Z. B. Simpson, M. Levy, and A. D. Ellington, Mol. Biosyst. 4(7), 754–761 (2008).
- ²⁶⁷C. M. Ajo-Franklin, D. A. Drubin, J. A. Eskin, E. P. S. Gee, D. Landgraf, I. Phillips, and P. A. Silver, Genes Dev. 21(18), 2271–2276 (2007).
- ²⁶⁸F. K. Balagaddé, H. Song, J. Ozaki, C. H. Collins, M. Barnet, F. H. Arnold, S. R. Quake, and L. You, Mol. Syst. Biol. 4, 187 (2008).
- ²⁶⁹A. S. Khalil and J. J. Collins, Nat. Rev. Genet. **11**(5), 367–379 (2010).
- 270 T. Kitada, B. DiAndreth, B. Teague, and R. Weiss, Science 359(6376), eaad1067 (2018).
- 271S. Slomovic, K. Pardee, and J. J. Collins, Proc. Natl. Acad. Sci. U. S. A. 112(47), 14429 (2015).
- **272**S. Huang and D. E. Ingber, Exp. Cell Res. **261**(1), 91–103 (2000).
- ²⁷³M. Xie and M. Fussenegger, Nat. Rev. Mol. Cell Biol. 19(8), 507–525 (2018).
- 274K. Roybal, L. Rupp, L. Morsut, W. Walker, K. McNally, J. Park, and W. Lim, Cell 164(4), 770–779 (2016).
- 275 N. E. Buchler, U. Gerland, and T. Hwa, Proc. Natl. Acad. Sci. U. S. A. 100(9), 5136–5141 (2003).

- 276 T. S. Moon, C. Lou, A. Tamsir, B. C. Stanton, and C. A. Voigt, Nature 491(7423), 249–253 (2012).
- ²⁷⁷K. T. Roybal, J. Z. Williams, L. Morsut, L. J. Rupp, I. Kolinko, J. H. Choe,
 W. J. Walker, K. A. McNally, and W. A. Lim, Cell 167(2), 419–432.E16
 (2016).
- **278**S. M. Douglas, I. Bachelet, and G. M. Church, Science **335**(6070), 831 (2012).
- ²⁷⁹X. Qu, S. Wang, Z. Ge, J. Wang, G. Yao, J. Li, X. Zuo, J. Shi, S. Song, L. Wang, L. Li, H. Pei, and C. Fan, J. Am. Chem. Soc. **139**(30), 10176–10179 (2017).
- 280 C. Zhang, W. Wang, N. Xi, Y. Wang, and L. Liu, Engineering 4(4), 452–463 (2018).
- ²⁸¹B. J. Williams, S. V. Anand, J. Rajagopalan, and M. T. A. Saif, Nat. Commun. 5(1), 3081 (2014).
- ²⁸²A. W. Feinberg, A. Feigel, S. S. Shevkoplyas, S. Sheehy, G. M. Whitesides, and K. K. Parker, <u>Science</u> **317**(5843), 1366 (2007).
- ²⁸³V. Chan, K. Park, M. B. Collens, H. Kong, T. A. Saif, and R. Bashir, Sci. Rep. 2(1), 857 (2012).
- ²⁸⁴J. C. Nawroth, H. Lee, A. W. Feinberg, C. M. Ripplinger, M. L. McCain, A. Grosberg, J. O. Dabiri, and K. K. Parker, Nat. Biotechnol. **30**(8), 792–797 (2012).
- ²⁸⁵S.-J. Park, M. Gazzola, K. S. Park, S. Park, V. D. Santo, E. L. Blevins, J. U. Lind, P. H. Campbell, S. Dauth, A. K. Capulli, F. S. Pasqualini, S. Ahn, A. Cho, H. Yuan, B. M. Maoz, R. Vijaykumar, J.-W. Choi, K. Deisseroth, G. V. Lauder, L. Mahadevan, and K. K. Parker, Science **353**(6295), 158 (2016).
- ²⁸⁶Y. Tanaka, K. Sato, T. Shimizu, M. Yamato, T. Okano, and T. Kitamori, Lab Chip 7(2), 207–212 (2007).
- ²⁸⁷J. Park, I. C. Kim, J. Baek, M. Cha, J. Kim, S. Park, J. Lee, and B. Kim, Lab Chip 7(10), 1367–1370 (2007).
- ²⁸⁸Y. Morimoto, H. Onoe, and S. Takeuchi, Sci. Rob. 3(18), eaat4440 (2018).
- ²⁸⁹S. Ostrovidov, V. Hosseini, S. Ahadian, T. Fujie, S. P. Parthiban, M. Ramalingam, H. Bae, H. Kaji, and A. Khademhosseini, Tissue Eng., Part B 20(5), 403–436 (2014).
- 290 A. Khodabukus and K. Baar, Cells Tissues Organs 202(3-4), 159-168 (2016).
- ²⁹¹Y. Akiyama, S.-J. Park, and S. Takayama, in *Nanotechnology and Microfluidics*, edited by X. Jiang, C. Bai, and M. Liu (Wiley-VCH, Weinheim, Germany, 2019), pp. 347–381.
- ²⁹²G. J. Pagan-Diaz, X. Zhang, L. Grant, Y. Kim, O. Aydin, C. Cvetkovic, E. Ko, E. Solomon, J. Hollis, H. Kong, T. Saif, M. Gazzola, and R. Bashir, Adv. Funct. Mater. 28(23), 1801145 (2018).
- ²⁹³H. Schmalbruch, Dev. Biol. **91**(2), 485–490 (1982).
- 294S. Balogh, C. C. Naus, and P. A. Merrifield, Dev. Biol. 155(2), 351-360 (1993).
- 295 R. Pfeifer, M. Lungarella, and F. Iida, Science 318(5853), 1088 (2007).
- ²⁹⁶S. Kriegman, D. Blackiston, M. Levin, and J. Bongard, Proc. Natl. Acad. Sci. U. S. A. **117**(4), 1853 (2020).
- ²⁹⁷P. Zimmet, K. G. Alberti, D. J. Magliano, and P. H. Bennett, Nat. Rev. Endocrinol. **12**(10), 616–622 (2016).
- ²⁹⁸F. Pampaloni, E. G. Reynaud, and E. H. K. Stelzer, Nat. Rev. Mol. Cell Biol. 8(10), 839–845 (2007).
- ²⁹⁹W. Li, S. Lee, M. Ma, S. M. Kim, P. Guye, J. R. Pancoast, D. G. Anderson, R. Weiss, R. T. Lee, and P. T. Hammond, Sci. Rep. 3(1), 2863 (2013).
- 300 A. Helman and D. A. Melton, Cold Spring Harbor Perspect. Biol. 13(1), a035741 (2021).
- ³⁰¹A. J. Vegas, O. Veiseh, M. Gürtler, J. R. Millman, F. W. Pagliuca, A. R. Bader, J. C. Doloff, J. Li, M. Chen, K. Olejnik, H. H. Tam, S. Jhunjhunwala, E. Langan, S. Aresta-Dasilva, S. Gandham, J. J. McGarrigle, M. A. Bochenek, J. Hollister-Lock, J. Oberholzer, D. L. Greiner, G. C. Weir, D. A. Melton, R. Langer, and D. G. Anderson, Nat. Med. 22(3), 306–311 (2016).
- 302 L. Schukur, B. Geering, G. Charpin-El Hamri, and M. Fussenegger, Sci. Transl. Med. 7(318), 318ra201 (2015).
- ³⁰³J. Livet, T. A. Weissman, H. Kang, R. W. Draft, J. Lu, R. A. Bennis, J. R. Sanes, and J. W. Lichtman, Nature **450**(7166), 56–62 (2007).
- 304S. Hampel, P. Chung, C. E. McKellar, D. Hall, L. L. Looger, and J. H. Simpson, Nat. Methods 8(3), 253–259 (2011).
- 305 D. Cai, K. B. Cohen, T. Luo, J. W. Lichtman, and J. R. Sanes, Nat. Methods 10(6), 540–547 (2013).
- 306 M. W. Veling, Y. Li, M. T. Veling, C. Litts, N. Michki, H. Liu, B. Ye, and D. Cai, Cell Rep. 29(10), 3303–3312.E3 (2019).

- ³⁰⁷Y. Li, L. A. Walker, Y. Zhao, E. M. Edwards, N. S. Michki, H. P. J. Cheng, M. Ghazzi, T. Y. Chen, M. Chen, D. H. Roossien, and D. Cai, "Bitbow: A digital format of Brainbow enables highly efficient neuronal lineage tracing and morphology reconstruction in single brains," bioRxiv:2020.04.07.030593 (2020).
- ³⁰⁸ A. Takai, M. Nakano, K. Saito, R. Haruno, T. M. Watanabe, T. Ohyanagi, T. Jin, Y. Okada, and T. Nagai, Proc. Natl. Acad. Sci. U. S. A. **112**(14), 4352 (2015).
- 309 J. Garcia-Marques, C.-P. Yang, I. Espinosa-Medina, K. Mok, M. Koyama, and T. Lee, Neuron 104(2), 227–238.e7 (2019).
- ³¹⁰ R. Sinha, C. M. Lacadie, R. T. Constable, and D. Seo, Proc. Natl. Acad. Sci. U. S. A. 113(31), 8837 (2016).
- ³¹¹D. Knox, B. R. Stanfield, J. M. Staib, N. P. David, S. M. Keller, and T. DePietro, Learn. Mem. **23**(12), 689 (2016).
- ³¹²E. Chukwurah, A. Osmundsen, S. W. Davis, and S. B. Lizarraga, Front. Neurosci. 13, 582 (2019).

- ³¹³C. A. Trujillo, R. Gao, P. D. Negraes, J. Gu, J. Buchanan, S. Preissl, A. Wang, W. Wu, G. G. Haddad, I. A. Chaim, A. Domissy, M. Vandenberghe, A. Devor, G. W. Yeo, B. Voytek, and A. R. Muotri, Cell Stem Cell 25(4), 558–569.E7 (2019).
- ³¹⁴M. Levin and C. J. Martyniuk, Biosystems **164**, 76–93 (2018).
- ³¹⁵D. M. Alon, C. A. Voigt, and J. Elbaz, ACS Synth. Biol. 9(2), 236–240 (2020).
- ³¹⁶J. Andersen, O. Revah, Y. Miura, N. Thom, N. D. Amin, K. W. Kelley, M. Singh, X. Chen, M. V. Thete, E. M. Walczak, H. Vogel, H. C. Fan, and S. P. Paşca, Cell 183(7), 1913–1929.E26 (2020).
- ³¹⁷I. van de Poel, in Handbook of Ethics, Values, and Technological Design: Sources, Theory, Values and Application Domains, edited by J. van den Hoven, P. E. Vermaas, and I. van de Poel (Springer, Dordrecht, Netherlands, 2015), pp. 667-690.
 ³¹⁸J. Stilgoe, R. Owen, and P. Macnaghten, Resour. Policy 42(9), 1568-1580
- ³¹⁸J. Stilgoe, R. Owen, and P. Macnaghten, Resour. Policy **42**(9), 1568–1580 (2013).
- ³¹⁹R. Raman, M. Mitchell, P. Perez-Pinera, R. Bashir, and L. DeStefano, J. Biol. Eng. **10**(1), 10 (2016).